Advances in Glucose Sensing Techniques: Novel Non-Invasive and Continuous Electrochemical Glucose Monitoring Systems

By

Kyoung Ha Cha

A dissertation submitted in partial fulfillment of the requirements for the degree of Doctor of Philosophy (Chemistry) in the University of Michigan 2018

Doctoral Committee:

Professor Mark E. Meyerhoff, Chair Assistant Professor Rodney C. Daniels Professor Raoul Kopelman Professor Kyung-Dall Lee Associate Professor Stephen Maldonado Kyoung Ha Cha

chakyoun@umich.edu

ORCID iD: 0000-0002-6898-251X

© Kyoung Ha Cha

DEDICATION

This work is dedicated to my family and my lovely wife.

ACKNOWLEDGEMENTS

First and foremost, I would like to express my sincere gratitude to my advisor and academic father, Professor Mark E. Meyerhoff, for his continuous support of my Ph.D. research, his patience, motivation, and immense knowledge. His guidance has helped me over many years of research and preparation of this doctoral dissertation. I cannot imagine a better advisor and mentor for my Ph.D. studies and life.

Aside from my advisor, I would also like to thank the rest of my thesis committee: Professors Raoul Kopelman, Stephen Maldonado, Kyung-Dall Lee, and Dr. Rodney C. Daniels, not only for their insightful comments and encouragement, but also for their hard questions which propelled me to widen my research from various perspectives. I would also like to express my sincere thanks to Dr. John Nicklas for giving me an opportunity to conduct a clinical study of red cell potassium measurements (not included in this dissertation) at a most prestigious institution, the Samuel and Jean Frankel Cardiovascular Center at the University of Michigan.

Further, I want to express my sincere thanks thanks to all the post-doctoral fellows and visiting scholars who provided a wealth of resourceful advice and wisdom during my Ph.D. studies; thank you Dr. Gary Jensen, Dr. Dipankar Koley, Dr. Yu Qin, Dr. Gergely Lautner, Dr. Woong Hee Lee, Dr. Xuewei Wang, Dr. Joanna Zajda, Dr. Kamila Konopinska, and Dr. Qi Zhang. I'd like to extend special thanks to Dr. Xuewei Wang, my lab mentor: Thank you for answering all of my questions and sharing with me your vast knowledge and experience.

I would also like to thank my fellow graduate students for many the stimulating discussions, encouragement and for all the good times we have had over the last four years;

thank you Dr. Andrea Bell, Dr. Elizabeth Brisbois, Dr. Wenyi Cai, Dr. Bo Peng, Dr. Si Yang, Dr. Alexander Wolf, Dr. Hang Ren, Dr. Alex Ketchum, Dr. Zheng Zheng, Dr. Yaqi Wo, Stephen Ferguson, and Joshua Doverspike for sharing my Ph.D. experience with you. I would also like to thank the undergraduates and visiting students that I've come to know during my Ph.D. work; thank you Anant Balijepalli, Maria Kim and Nicholas Schott.

And finally, last but by no means least, I would like to thank my family, especially my parents, Dr. Geun-Sig Cha and Hyun Sook Eo, for your unconditional love, support and trust. I would also like to thank my wife, Christine Kichung Byun, for the same unconditional support. In addition, I want to thank my sister, Eunice Cha, and her husband, Keith Ryoo, for their continuous love and support.

TABLE OF CONTENTS

DEDICATION	ii
ACKNOWLEDGEMENTS	iii
LIST OF FIGURES	vii
LIST OF TABLES	x
ABSTRACT	xi
CHAPTER 1. Introduction	1
1.1 Current Status of Diabete Mellitus	1
1.2 Non-Invasive Glucose Monitoring Systems	2
1.3 Nitric Oxide Release for Improving Performance of Implantable Chemical Sensors	3
1.3.1 Challenge 1.3.2 Nitric oxide	5 6
1 3 3 NO-releasing chemical sensors	0
1 4 Summary	20
1.5 Statement of Research	
1.6 References	23
CHAPTER 2 Evaluation of Commercial Clucometer Test Strips for Potential Mass	uromont
of Clucose in Tears	27
2 1 Introduction	·····27
2 2 Experimental Section	30
2.2.1 Chemicals and reagents	
2.2.2 Sensitivity and selectivity analysis	
2.2.3 Human tear sample aquisition	
2.2.4 Human tear sample glucose measurement	
2.3 Results and Discussion	
2.3.1 Sensitivity analysis of glucometer strips	
2.3.2 Selectivity analysis of glucometer strips	
2.3.3 Selectivity analysis for different lots of glucometer strips	
2.3.4 Preliminary assessment of Roche strips for measurements of glucose in human tears	
2.4 Conclusions	
2.5 References	
CHAPTER 3. Origin of Low Detection and High Selectivity of Roche Accu-Chek Te	est Strips
that Enables Measurement of Tear Glucose Levels	
3.1 Introduction	45
3.2 Experimental Section	47
3.2.1 Chemicals and reagents	47
3.2.2 Electrochemical measurements	
3.3 Results and Discussion.	
3.3.1 Sensor configuration for tear glucose measurement	

3.3.2 Selectivity analysis	51
3.3.3 Sensitivity analysis	54
3.3.4 Optimization for measurement of low-end glucose level	58
3.4 Conclusions	60
3.5 References	61
CHAPTER A Determination of Correlation Retwoon Toor and Blood Clucose Level	s of
Human Subjects with Type 2 Diabetes Using Acou Chely Test String	5 UI 62
A 1 Introduction	62
4.1 Introduction	63
4.2 Experimental Section	63
1.2.1 Materials	63
4.2.2 Sensitivity and selectivity analysis	63
4.2.5 Human tear sample acquisition and Analysis	05 64
4.5. Results and Discussion.	67
1.5 References	07
	00
CHAPTER 5. Compatibility of Nitric Oxide Release with Implantable Enzymatic Gluo	cose
Sensors Based on Osmium (III/II) Mediated Electrochemistry	. 69
5.1 Introduction	69
5.2 Experimental Section	72
5.2.1 Sensor fabrication	72
5.2.2 Sensor performance	73
5.3 Results and discussion	74
5.3.1 Effect of nitric oxide on osmium-mediated chemistry	74
5.3.2 Nitric oxide flux analysis	76
5.3.3 Detection of copper leaching	77
5.3.4 Sensor performance on benchtop	78
5.3.5 Improving sensors to reduce oxygen sensitivity	80
5.3.6 Sensor performance in vivo	81
5.4 Conclusions	83
5.5 References	84
CHAPTER 6. Conclusions and Future Directions	86
6.1 Conclusions	86
6.2 Future Directions	89
6.2.1 Tear fluid acquisition method for tear glucose measurement	89
6.2.2 Development of combined insulin cannula with osmium-mediated glucose sensor.	90
6.2.3 Development of NO-releasing 3 rd generation glucose sensor	93
6.3 References	94

LIST OF FIGURES

Figure 1.1. Sequential process that ultimately leads to thrombus formation on the foreign device Figure 1.2. The structures of some NO donors/generators: (a) (Z)-1-[N-butyl-N-[6-(Nmethylammoniohexyl)amino]]-diazen-1-ium-1,2-diolate (DBHD/N₂O₂), (b) S-nitroso-N-acetyl-D-penicillamine (SNAP), and (c) Copper(II)-tri(2-pyridylmethyl)amine (Cu(II)TPMA), a catalyst Figure 1.3. (a) NO-releasing needle/catheter type glucose sensor with DBHD/N₂O₂ as NO donor; (b) photos of the control (top) and NO release (bottom) glucose sensors after the in vivo experiment. The portions of the sensors to the left of the dashed lines were actually inside the veins Figure 1.4. (a) Schematic of dual-lumen catheter-type electrochemical NO generating/releasing PO_2 sensor with cross-section geometry of catheter; (b) representative photo illustrating the degree of clot formation on the surface of the control and the NO releasing sensors implanted in rabbit Figure 1.5. (a) Designs of implantable NO-releasing osmium-mediated glucose sensor; (b-d) in vitro comparison of intermittent glucose values obtained from a commercial glucometer and continuous glucose measurements obtained from a NO-releasing osmium-mediated glucose sensor Figure 2.1. Calibration data for 0, 50, and 100 µM standard glucose solution using 10 s average current recorded starting at the 60-s time point after sample introduction (at 37 °C, applied voltage to working electrode = +400 mV) for several different commercial glucometer strips: (A) FreeStyle Lite[®] by Abbott; (B) Precision Xtra[®] by Abbott; (C) CONTOUR[®] by Bayer; and (D) OneTouch[®] Ultra® Blue by Johnson & Johnson. Error bars are standard deviations (S.D.) for n = 5Figure 2.2. (A) Dynamic amperometric response (at 37 °C; applied potential = + 400 mV, LOO = 44 μ M) and (B) resulting calibration data for ACCU-CHEK[®] Aviva Plus strips (Roche Diagnostics) with PBS solution (pH 7.4, 0 µM glucose, no interference) (black) and glucose concentrations of 50 (red), 100 (blue) µM) with the 10 s average current recorded 5 s following Figure 2.3. Calibration data for 0-100 µM glucose solutions with average currents over 10 s period recorded beginning 5 s after sample introduction (at RT) using (A) ACCU-CHEK[®] Aviva Plus strips and (B) TRUEtest® strips at different applied voltages. All test solutions contained 10 µM acetaminophen, 100 µM ascorbic acid, and 100 µM uric acid. Both sets of calibration data are Figure 2.4. Dynamic amperometric response to ACCU-CHEK® Aviva Plus strips with PBS solution (pH 7.4, 0 µM glucose, no interference) (black) and of glucose at 25 (red), 50 (blue), 75

(green) µM with electroactive interferences present (10 µM acetaminophen, 100 µM ascorbic acid, Figure 3.1. Structures of (a) ferricyanide and (b) O-methoxy-[N,N-bis-(2-hydroxyethyl)]-p-Figure 3.2. Schematic of blood glucometer test strip and its potential application for monitoring glucose levels in a tiny volume of tear fluid. In addition to gold and carbon electrodes utilized in Figure 3.3. Comparison of dynamic amperometric responses (at room temperature; applied potential = +150 mV) from the interferents (0.5 mM ascorbic acid (AA), 0.5 mM uric acid (UA) and 0.2 mM acetaminophen (AM)) in the presence of 10 mM potassium ferricyanide (a) or 10 mM Figure 3.4. Comparison of dynamic amperometric reponses (at room temperature; applied potential = +150 mV) from the interferents (0.5 mM ascorbic acid (blue), 0.5 mM uric acid (red) and 0.2 mM acetaminophen (green)) in the presence of 10 mM potassium ferricyanide (a, c) or 10 Figure 3.5. Dynamic amperometric response (at room temperature; applied potential = +150 mV) for changes in glucose concentration (0-5 mM) in the presence of (a) 20 mM potassium ferricyanide and (b) 20 mM nitrosoaniline derivative in PBS solution (pH 7.4) with 125 U/ml glucose oxidase (GOX), FAD- (FAD-GDH), and PQQ-dependent glucose dehydrogenase (PQQ-Figure 3.6. Comparison of dynamic amperometric responses (at room temperature; applied potential = +150 mV vs. Ag/AgCl in the presence of 10 mM potassium ferricyanide and 10 mM potassium ferricyanide before (solid lines) and after (dashed lines) measuring 20 mM Figure 3.7. Dynamic amperometric response (at room temperature; applied potential = +150 mVvs. Ag/AgCl) for increasing glucose concentrations (0-1 mM) in the presence of 10 and 3.7 mM nitrosoaniline derivative (ND), PQQ-dependent glucose dehydrogenase with various Figure 4.1. Calibration data for glucose using 2 s average current recorded starting at the 2 s time point after sample introduction with an applied potential of +150 mV vs. Ag/AgCl. (a) 0-100 μ M glucose calibration obtained by using test strips (Lot No. 496722). The limit of detection (LOD) Figure 4.2. Chronological changes in tear and blood glucose levels of n = 4 human subjects with type 2 diabetes. The collection order at every 0.5 hr is as following: right eye, left eye and blood (drawn from lancet); Glucose challenge was introduced right after the first set of collection at 0 hr Figure 4.3. Correlation between tear and blood glucose levels in n = 4 human subjects with type Figure 5.1. Designs of implantable (a) osmium-mediated glucose sensor and (b) NO-releasing Figure 5.2. Amperometric response to glucose (0-20 mM) obtained with non-NO release Os(III/II)-mediated implantable glucose sensor (Fig. 5.1a) with applied potential of +200 mV vs. Ag/AgCl on days 1-5.....74 Figure 5.3. A comparison of continuous current measurements for 0 mM glucose using a glucose sensor based on hydrogen peroxide detection (+650 mV vs. Ag/AgCl) and osmium-mediated chemistry (+200 mV) in the absence and presence of 300 ppm NO purged into the solution at the

10 min mark. Current was normalized by subtracting the background current at 0 mM glucose for Figure 5.4. Continuous measurement of 0 mM glucose (0-3 h) and 14 mM glucose (3-25 h) using osmium-mediated glucose sensor with applied potential of +200 mV vs. Ag/AgCl. Gas phase NO at 300 ppm in nitrogen was purged between 1-25 h. Inset: Corresponding calibration curve of Figure 5.5. NO flux analysis of SNAP-based NO release glucose sensors prepared with and without Cu-NPs for 3 consecutive days (note: 1 flux unit = $1 \times 10^{-10} \text{ mol·min}^{-1} \cdot \text{cm}^{-2}$). The error Figure 5.6. Glucose calibrations (0-20 mM) with applied potential of 200 mV in PBS (pH 7.4, 37 °C) for 3 consecutive days. (a) NO-releasing glucose sensor; (b) control glucose sensor. (note:1 Figure 5.7. In vitro comparison of intermittent glucose values obtained from a commercial glucometer and continuous glucose measurements obtained from a NO-releasing osmiummediated glucose sensor as well as a control (no NO release) osmium-mediated glucose sensor poised at 200 mV in heparinized whole porcine blood for 3 days. On each day, glucose values Figure 5.8. (Left) Glucose calibrations (0-20 mM) and obtained by new osmium-mediated glucose sensors (n=4) that utilize FAD-GDH as glucose-oxidizing enzyme with applied potential of +200 mV vs. Ag/AgCl in PBS (pH 7.4, 37 °C). (Right) Continuous measurement of 20 mM glucose (0-24 h) using the new osmium-mediated glucose sensors (n=3) with applied potential of +200 mV Figure 5.9. Performance of electrochemical NO-releasing osmium-mediated glucose sensors Figure 6.1. Potential design of tear glucose sensor featuring a soft housing for holding the Figure 6.2. Glucose-sensing cannula results obtained in anesthetized nondiabetic swine using both H₂O₂-detection electrode (upper figure) and redox mediator sensor (lower figure). Arrows at 0 min time-marks in both upper and lower figures represent the timing of lis-pro insulin delivery. Note: Figure 6.3. Calibration curves obtained using glucose sensors based on H₂O₂-detection (left) and NO fluxes (right) of the integrated device in a 18-day experiment in phosphate buffer solution. 92 Figure 6.4. (a) Calibration of 3rd generation glucose sensor utilizing CDH in TRIS buffer (pH 7.4) with an applied potential of +0.25 V vs. Ag/AgCl. Inset: Linear part of the calibration curve....93

LIST OF TABLES

Table 2.1. Enzyme and mediator combinations in commercial glucometer strips and preliminary
assessment of detection capability at low level glucose concentrations
Table 2.2. Selectivity test results for ACCU-CHEK® Aviva Plus and TRUEtest® strips
Table 2.3. Selectivity test results for six lots of ACCU-CHEK® Aviva Plus strips at 150 mV40
Table 2.4. Summary of fasting human tear glucose measurements with ACCU-CHEK® Aviva Plus
strips for 9 non-diabetic volunteers
Table 2.5. Summary of fasting human tear glucose measurements with ACCU-CHEK® Aviva Plus
strips and LC-ESI-MS from Chapter 2, Ref. 23
Table 5.1. Copper leachate concentration from Cu-NPs in Carbosil layer within osmium-mediated
glucose sensor and relative leaching into PBS as compared to the original amount of copper present
in each sensor

ABSTRACT

Diabetes mellitus (or diabetes) is a chronic, lifelong condition that affects the body's ability to utilize the stored chemical potential energy found in our food. Frequent measurement and tight control of blood glucose is essential to avoiding life-threatening hyper- and hypoglycemic events and associated serious, long-term complications. In this dissertation, a novel non-invasive tear glucose measurement approach and various continuous electrochemical glucose sensor-based monitoring devices with nitric oxide (NO) release are examined and evaluated for their potential application for diabetes management. Tear glucose measurements have been previously suggested as a potential alternative to blood glucose monitoring for diabetic patients, although this approach has not been thoroughly established. In Chapter 2, the first use of commercial blood glucose test strips to measure glucose in tears is examined. Roche AccuChek test strips are shown to exhibit the low detection limit required for quantitating glucose concentration in tears. Measurements of glucose in tears from nine normal (nondiabetic) fasting human subjects using strips yielded glucose values within the range of $5-148 \mu$ M, similar to glucose measurements for human tears reported by others via LC-MS methods. Chapter 3 evaluates the origin of the high sensitivity and selectivity of the Roche test strips and demonstrates that the use of pyrrologuinoline guinone (PQQ)dependent glucose dehydrogenase (GDH) in combination with a nitrosoaniline derivative as an electron transfer mediator provides the low limits of quantitation (ca. 9 μ M) and enhanced selectivity achieved with these strips. In Chapter 4, the test strips are used to measure glucose

levels in tear fluids from human subjects with type 2 diabetes after fasting and then for 90 min after ingesting sugar while concurrently measuring the blood glucose values. A moderate correlation between tear and blood glucose levels is demonstrated.

Tight glycemic control helps reduce life-threatening hyper- and hypoglycemic events that can cause serious long-term complications for hospitalized critically ill patients. Therefore, the development of continuous glucose monitoring systems to quantitate blood glucose levels intravascularly (IV) could improve patient outcomes. In Chapter 5, the compatibility of nitric oxide (NO) release coatings with implantable enzymatic glucose sensors based on osmium (III/II) mediated electrochemical detection is examined for the first time. Nitric oxide (NO) is a potent inhibitor of platelet activation and adhesion. NO-releasing osmium-mediated glucose sensors are prepared using a *S*-nitrosothiol impregnated outer tubing and are tested *in vitro* in both phosphate buffer (pH 7.4) and heparinized whole porcine blood. After 3 days of continuous NO release at or above physiological levels, no negative effects on the osmium mediated electrochemical currents are observed. These results suggest that improved performance of both intravascular and, potentially, subcutaneous Os(III/II) mediated glucose sensors may be realized by incorporating NO's well-known anticlotting, anti-inflammatory, and antimicrobial properties.

CHAPTER 1

Introduction

1.1 Current Status of Diabetes Mellitus

The World Health Organization (WHO) reports that more than 347 million people worldwide live with diabetes. Over 1.1 million deaths in 2005 were caused by diabetes, and deaths attributed to diabetes are estimated to double by 2030.¹ Typically, a personal blood glucometer with single-use test strips allow diabetic patients to monitor their blood glucose level by obtaining and quantitating a small sample of capillary blood with a lancet, either from the fingertip or forearm. Optimal metabolic control is possible for these patients with a minimum of four blood glucose checks per day.³ However, for children and adolescents with Type 1 diabetes undergoing insulin therapy, this method of glucose monitoring is often recommended up to eight times daily.⁴ The resulting discomfort limits patient compliance and leads to suboptimal blood glucose control as well as the risk of hypoglycemic incidents. The development of a pain-free, non-invasive sensing technique would greatly improve patient compliance and, ultimately, patient outcomes. At the same time, the ideal technology is to continuously monitor glucose levels in blood with an implantable sensor. This thesis focuses on developing/evaluating new approaches/devices that contribute to both goals. A major portion of this chapter was published as a review paper in *Applied Materials Today*.⁵

1.2 Non-Invasive Glucose Monitoring Systems

Many studies have aimed to establish a useful yet noninvasive technique for monitoring blood glucose levels. The methods applied include infrared (IR) spectroscopy,⁶⁻⁷ Raman spectroscopy,⁸⁻⁹ optical coherence tomography, measurement of tissue metabolic heat conformation¹⁰ and electrochemical analysis of sweat;¹¹ however, none of these methods have proven to be clinically applicable.¹² In addition, several *in vivo* sensing approaches have been proposed, including fluorescence spectroscopy,¹³⁻¹⁴ surface plasmon resonance of nanoparticles,¹⁵⁻¹⁶ electrical impedance measurements, and implantable/subcutaneous amperometric glucose sensors. Among these, only subcutaneous electrochemical glucose sensors are available commercially, although frequent calibration, short *in vivo* lifetimes, limited accuracy, and high cost have limited their widespread use.¹⁷ Further, such sensors are clearly much more invasive.

Measurement of tear glucose concentrations is an emerging area of research and several studies have reported a possible clinically useful correlation between blood and tear glucose levels.¹⁸⁻²¹ Beca *et al.* reviewed studies of this correlation using different detection methods,²² concluding that there is evidence of a correlation. Methods for tear glucose measurement require a low micromolar (μ M) range for the limit of quantitation (LOQ), high selectivity over potential interferences (e.g., ascorbic acid and uric acid), and measurement within μ L sample volumes. To date, tear glucose has been measured by capillary electrophoresis (CE) coupled with laser-induced fluorescence (LIF),²³ fluorescence sensors,²⁴⁻²⁵ liquid chromatography (LC) coupled with electrospray ionization mass spectrometry (ESI-MS),²⁶ holographic glucose sensors,²⁷ a miniaturized flexible thick-film flow-cell electrochemical detector,²⁸ a strip-type flexible biosensor²⁹ and an amperometric biosensor with microflow injection analysis.³⁰ Badugu et al. and Ruan et al. also reported the feasibility of using disposable contact lenses to monitor tear glucose.³¹⁻³³ A study from the Asher laboratory²⁶ employed a LC-ESI-MS technique to measure glucose concentrations in 1 μ L tear samples obtained from non-diabetic subjects and reported a median value of 28 μ M with a range from 7-161 μ M for this group.

1.3 Nitric Oxide Release for Improving Performance of More Invasive Implantable Chemical Sensors

1.3.1 Challenge

The real-time and accurate measurements of clinically important species, such as arterial blood gases (pH, *PO*₂, *PCO*₂), electrolytes (Na⁺, K⁺, Ca²⁺) and glucose/lactate levels in undiluted circulating whole blood of hospitalized patients via implantable chemical sensors continues to be of great interest to improve the quality of health care.³⁴⁻³⁶ This is because intermittent *in vitro* blood measurements cannot provide real-time information on changing physiological conditions of critically ill patients, which can lead to life-threatening events. However, designing successful *in vivo* sensors with sufficient accuracy for clinical use is complicated by adverse biological responses that lead to errant analytical results.³⁷⁻³⁹

The main biological response to implanted sensors depends largely on the location of implantation: blood or subcutaneous tissue.⁴⁰ When an implantable chemical sensor is in contact with flowing whole blood, the initial biological response is the nearly instant adsorption of plasma proteins (e.g., fibrinogen, fibronectin, vitronectin and von Willebrand's factor (vWF)) onto the foreign device surface, which leads to the adhesion, activation and aggregation of platelets that trap red blood cells and ultimately forms a thrombus (blood clot) within a matter of hours (see Fig. 1.1).⁴¹⁻⁴⁴ Such biofouling on the surface of implantable sensors can restrict the mass transport of analytes to the



Figure 1.1. Sequential processes that ultimately lead to thrombus formation on the foreign device surface.

sensing surface, leading to deterioration in sensor response/performance.⁴⁵⁻⁴⁶ Further, the metabolic activities of adhered cells consume oxygen and glucose while producing carbon dioxide, thereby altering the local concentration of these species measured by the implanted chemical sensors, leading to inaccurate results.³⁹

Sensors implanted in the subcutaneous tissue (under the skin) are also subject to biocompatibility problems, primarily from inflammatory responses by the body's self-defense mechanism which can result in inaccurate results.^{37,46} This acute inflammatory response rapidly takes place after sensor implantation as proteins and inflammatory cells migrate to the implant site and adhere to

the sensor surface. Subsequently, local edema is created at the surface. Further, phagocytic cells (e.g., neutrophils, monocytes, and macrophages) form fibrous encapsulation around the sensors, creating a significant change of the analyte diffusion rate, resulting in frequent changes in the response curve/sensitivity of the sensor.^{47.49} In addition, the injury to blood vessels caused by implanting chemical sensors in the subcutaneous tissue will facilitate the diffusion of the analyte from the blood to the subcutaneous fluids affecting the output accuracy of sensors.⁵⁰ To date, attempts to monitor in subcutaneous tissue have mostly been focused on glucose measurements, since the concentrations of glucose in this tissue closely relate to blood levels.⁵¹⁻⁵² The critical parameters for animal models and data processing methods required to assess the reliability and usefulness of such subcutaneous glucose sensors have also been studied.⁵³

Many strategies have been reported to reduce biofouling on the surface of implanted chemical sensors by using various types of modified polymers.⁵⁴⁻⁵⁹ Some examples include hydrogels,^{44,47} phospholipid-based biomimicry, flow-based systems, Nafion polymer coatings, surfactant-derived membranes, diamond-like carbons, use of polyurethane and silicone elastomers, and topology treatments.^{45,60-67} Also, many efforts have been made to design "active" biocompatible materials, most often by immobilization of heparin, to prevent significant thrombus formation at the surface of intravascular devices.⁶⁸⁻⁷¹ However, platelet cell adhesion and activation can still occur, leading to localized changes in analyte levels. Furthermore, the use of immobilized heparin may be problematic for patients who suffer from thrombocytopenia, hemophilia or renal failure. Despite such extensive efforts, none of these methods have proven to be useful for fabrication of clinically effective implantable chemical sensors for use in monitoring critically ill hospital patients.

1.3.2 Nitric Oxide

Nitric oxide (NO) is a diatomic free radical and endogenously produced from L-arginine and oxygen via nitric oxide synthase (NOS) enzyme within the endothelial cells that line the inner walls of all blood vessels.⁷² Nitric oxide is known to prevent platelet activation/adhesion and subsequent thrombosis, as well as serving as an inhibitor of bacterial cell proliferation and biofilm formation.⁷³⁻⁷⁶ Indeed, macrophage cells in our bodies, which fight infection, produce high levels of NO for this purpose. Further, it is known that NO at low levels is also a potent anti-inflammatory agent.⁷⁷ Hence, the prevention of thrombosis, infection, and foreign body response associated with implantable chemical sensors may be possible by incorporating them within platforms capable of releasing low levels of NO at physiological fluxes of ca. $0.5-4.0 \times 10^{-10}$ mol·cm⁻²·min⁻¹.⁷⁸ Nitric oxide concentrations can be most accurately measured via a chemiluminescence method.⁷⁹

It is also widely known that NO can be highly reactive with *in vivo* species (oxyhemoglobin, oxygen, and thiols). Indeed, the life-time of NO in blood is very short, $<1 \text{ s.}^{43}$ Nonetheless, its lifetime is still long enough to have important physiological activities at contact surfaces, especially serving to inhibit platelet activation (preventing clotting when produced at flux rates that mimic those produced by endothelial cells). NO may also be considered as a potential interference for *in vivo* oxygen sensing. However, Ren et al. showed that an implantaable amperometric O₂ sensor is fully compatible with NO release, as no noticeable amperometric signal changes were observed for the O₂ sensor with NO generation at physiological flux levels.⁸⁰ Further, Nichols et al. evaluated the *in vivo* glucose recovery of subcutaneously implanted NO-releasing (from saturated NO solution) microdialysis probes in a rat model over 14 d period and reported that intermittent sustained NO release from implanted probe surfaces may improve

glucose diffusion for such subcutaneously implanted devices by mitigating the foreign body reaction.⁸¹ However, since NO has a short half-life, many NO donors have been studied for achieving prolonged and controlled NO delivery for potential clinical use, instead of using NO gas from a cylinder.

1.3.3 NO-releasing Chemical Sensors

1.3.3.1 N-Diazeniumdiolates as NO donors

A N-diazeniumdiolate is an NO adduct with secondary amines,⁸² and one of the most widely investigated NO donors due to its ability to release NO spontaneously under physiological conditions via a proton-driven reaction.⁸³⁻⁸⁴ One equivalent of amine reacts with two equivalents of NO to form the corresponding diazeniumdiolate under high NO(g) pressure (e.g., 80 psi). An initial study by Espadas-Torre et al. reported the fabrication of classical polymer membrane type ion-selective electrodes (ISE) for H⁺ and K⁺ with a NO donor (N, N'-dimethylhexanediamine nitric oxide adduct, also known as $DMHD/N_2O_2$) within to investigate whether low level of continuous NO release from the diazeniumdiolate species would impair the analytical performance of the sensors.⁸⁵ The results showed that ISE sensors with and without DMHD/N₂O₂ exhibited very comparable potentiometric responses: K⁺-sensors with and without the added diazeniumdiolate species (membrane composition: 66% Tecoflex polyurethane/30% dioctyl sebacate/0.6% potassium tertakis(4-chlorophenyl)borate/2.5% valinomycin) exhibited slopes of 60.3 ± 0.1 and 60.6 ± 0.2 mV/decade and intercepts of -5.31 and -5.44, respectively. For H⁺-detection (membrane composition: 33% Tecoflex polyurethane/63% dioctyl sebacate/1.7% potassium tertakis(4-chlorophenyl)borate/2.5% tridodecylamine) the sensors exhibited slopes of 58.5 ± 0.9 and 59.7 ± 0.6 mV/decade with intercepts of -9.38 and -9.50, respectively. This study also

reported that the surfaces of the sensors exhibit a marked decrease in thrombogenic properties when measured by *in vitro* platelet adhesion studies using sheep plasma. This early study clearly suggested that their approach for enhancing biocompatibility of ion sensors by incorporation of a NO donor can be applicable to a wide range of polymeric materials that can be used in implantable chemical devices. In another early study, Mowery et al. prepared Clark style amperometric oxygen sensors and potentiometric sensors (pH and K⁺) with three distinctly different types of diazeniumdiolate NO donors, DMHD/N₂O₂, linear polyethylenimine/N₂O₂ (LPEI/N₂O₂) and methoxymethyl piperazine PVC/N₂O₂ (mompipPVC/N₂O₂), and reported a significant decrease in platelet adhesion and activation for both *in vitro* and *in vivo* sensors fabricated with these materials.⁸⁶

The effect of diazeniumdiolate species was further investigated for potential improvement of the biocompatibility and *in vivo* analytical performance of intravascular amperometric oxygen sensors by Schoenfisch et al. In this work, the authors fabricated NO-releasing silicone rubber-catheter-based amperometric *P*O₂ sensors via dip-coating the sensor with gas permeable coatings formulated with cross-linked silicone rubber containing DMHD/N₂O₂. They then demonstrated that there was greatly reduced thrombus formation on the outer surface of the sensors and superior overall sensor performance vs. control sensors when implanted, in the absence of systemic anticoagulation, within the carotid and femoral arteries of mongrel dogs for 6-24 h.⁸⁷ The only caveat was leaching of N,N'dimethyl-hexane-diamine and the corresponding toxic nitrosamine, a decomposition product of DMHD/N₂O₂, into aqueous soaking solutions or blood. Therefore, they concluded that NO donors such as diazeniumdiolates that possess a more lipophilic structure, or can be covalently attached to the outer polymer coating of the sensor would be more optimal.

Frost et al. decreased the potential problem of leaching diamine and nitrosoamine by using a more lipophilic NO-releasing agent within the silicone rubber polymer that contains a diazeniumdiolate species (DACA-6/N₂O₂), anchored to the cross-link sites of a silicone rubber matrix.⁸⁸ They further prepared Clarke-style amperometric oxygen-sensing catheters coated with an outer layer of the DACA-6/N₂O₂ polymer that releases physiological levels of NO for >20 h, and showed significantly improved biocompatibility and analytical performances compared to control sensors without NO release when tested *in vivo* within the carotid and femoral arteries of swine for 16 h.

By coating a dual lumen silicone catheter with this DACA-6/N₂O₂-silicone layer, an NO release potentiometric CO₂ sensor was also developed.⁸⁹⁻⁹⁰ The wall between two lumens was doped with a proton ionophore, tridodecylamine, and a tetraphenylborate type cation exchanger, functioning as a pH-selective layer. One lumen contained a bicarbonate/sodium chloride solution and a Ag/AgCl electrode. The pH of this lumen solution decreased in the presence of an increase in the *P*CO₂ level in solution (or blood) surrounding the device. This causes a potential change across the polymeric wall between the lumen filled with the bicarbonate solution and the other lumen filled with strong buffer (4-morpholinoethanesulfonic acid buffer, pH = 5.7). A second Ag/AgCl reference electrode in the buffered lumen provides a means to measure this potential change as a function of change in *P*CO₂ levels. The overall potential change of this two-electrode system was found to be proportional to log(*P*CO₂) of the sample with a near-Nernstian slope (-60.1 to -57.8 mV/decade) even in the presence of ~ 10 × 10⁻¹⁰ mol·cm⁻²·min⁻¹ fluxes of NO from the surface of the device.



Figure 1.2. The structures of some NO donors/generators: (a) (Z)-1-[N-butyl-N-[6-(N-methylammoniohexyl)amino]]-diazen-1-ium-1,2-diolate (DBHD/N2O2), (b) *S*-nitroso-*N*-acetyl-D-penicillamine (SNAP), and (c) Copper(II)-tri(2-pyridylmethyl)amine (Cu(II)TPMA), a catalyst used for electrochemical generation of NO from an inorganic nitrite reservoir.

Given that the NO-releasing implantable PO₂ sensors showed promising in vivo results, Gifford et al. developed the very first NO-releasing needle-type enzyme-based electrochemical glucose sensors for measurement of glucose in subcutaneous fluid.⁹¹ The outer membrane of this glucose sensor contained а diazeniumdiolate species, (Z)-1-[N-methyl-N-[6-(Nbutylammoniohexyl)amino]]-diazen1ium-1,2-diolate (DBHD/N₂O₂; see Fig. 1.2 for chemical structure), that can release NO for ~ 16 h. The resulting sensor showed excellent analytical performance for both in vitro and in vivo tests and inhibited the typical subcutaneous inflammatory response when implanted in Sprague-Dawley rats for 3 d. Clarke error grid analysis indicated superior performance for the NO-releasing sensors when compared to controls on both days 1 and 3. Histological examination of the sites where the NO-releasing sensors were placed in the animals also showed significantly less inflammatory response over the first 24 h compared to non-NO



Figure 1.3. (a) NO-releasing needle/catheter type glucose sensor with DBHD/N₂O₂ as NO donor; (b) photos of the control (top) and NO release (bottom) glucose sensors after the *in vivo* experiment. The portions of the sensors to the left of the dashed lines were actually located inside the veins; (c) Comparison of glucose concentration values obtained from benchtop blood gas analyzer and the converted current values measured by the continuous sensor. One conversion of current to glucose concentration (mmol·L⁻¹) was made with the calibration curve in bovine serum; the other conversion was a one-point calibration. Adapted from Ref. 93.

release controls, and a significant decrease in the *in vivo* sensor signal stabilization time right after implantation. Yan et al. and Wolf et al. also developed the same type of implantable amperometric glucose sensors based on hydrogen peroxide detection with an outer polymer containing NO releasing DBHD/N₂O₂ and found reduced thrombus formation and improved analytical

performance compared to control sensors when implanted in the veins of rabbits for 7 h (see Fig. 1.3).⁹²⁻⁹⁴ However, these studies found that NO release from a thin layer of DBHD/N₂O₂ was too short and uncontrollable for ultimate clinical application and that NO could also be detected as an interference current signal at an applied potential of +0.6 V vs. Ag/AgCl and higher on the underlying platinum–iridium working electrode, which was required for electrochemical glucose sensor measurement via detection of hydrogen peroxide produced from glucose oxidase. Overall, these studies indicated that NO is compatible with subcutaneously and intravascularly-implanted electrochemical glucose sensors, but additional work on the best NO release chemistry is still necessary.

The need for improved NO release coatings to enhance the *in vivo* performance of implantable chemical sensors led to the development of an NO-releasing hydrophilic platform. Marxer et al. first developed an amperometric sol-gel derived oxygen sensor that releases NO via coating the platinum working electrode with an aminosilane/ethyltrimethoxysilane hybrid xerogel film doped with hydrophilic polyurethane (HPU) that are subsequently exposed to 5 atm of NO for 3 d (to form diazeniumdiolate groups on the amine sites).⁹⁵ The sensors exhibited highly reproducible and linear sensitivities to oxygen with rapid response times and released NO at fluxes of 7.2×10^{-10} mol·cm⁻²·min⁻¹, effectively reducing platelet adhesion for 48 h. However, it was suggested that further work to improve the NO release layer was needed to evaluate the *in vivo* performance of such implantable chemical sensors. Indeed, Shin et al. developed a macro glucose sensor based on a new type of hybrid sol-gel/polyurethane (PU), made by sequentially coating the platinum working electrode with the following four polymeric membranes: a sol-gel layer with immobilized glucose oxidase (GOX), PU for protection of the GOX, PU doped with NO donor

(diazeniumdiolated sol-gel particles) and sol-gel particles, and then PU as an outermost layer.⁹⁴ The NO donor was prepared by exposing aminosilane-based sol-gel particles to 5 atm NO for 3 d to convert accessible amine groups to diazeniumdiolates. The design demonstrated good sensitivity, reproducibility, and fast response time to glucose as well as a reduction of enzyme inactivation. However, the authors observed leaching of sol-gel particles even when the sensor was coated with an outermost PU layer and a drastic decrease in sensitivity of sensor to glucose when coupled with this new NO-releasing sol-gel film. Also, a longer duration of higher NO release fluxes was still required for the miniaturized version of sensor to study the in vivo biocompatibility of such a design. Indeed, Oh et al. reported a miniaturized glucose biosensor modified with a NO-releasing xerogel micro-patterned array that released NO from 5 µm wide lines separated by 5 or 20 µm distances.⁹⁶ The micro-patterned array allowed for an increase in glucose sensitivity since significant portions of the sensor's surface remained unmodified by the NO release coating. They found that sufficient levels of NO generation from such a design significantly reduced platelet adhesion (>40% reduction) and *Pseudomonas aeruginosa* bacterial growth (70-80% reduction) without reducing enzymatic activity of GOX. However, since the NO release from sensor depleted after only 3 days, extension of NO release time would be necessary for longer-term *in vivo* testing of such miniaturized glucose biosensors.

Koh et al. further investigated the fabrication of polyurethane-based implantable glucose sensors with a NO-releasing porous fiber mat-modified sensor membrane (540 ± 139 nm fiber diameter, $94.1 \pm 3.7\%$ porosity).⁹⁷ The sensor released ~100 nmol of NO per mg of polyurethane over 6 h period without leaching of the NO donor, even in serum. However, to date, this new design has never been evaluated for *in vivo* analytical performance.

A diazeniumdiolate type-NO donor was also used to provide NO release for film type fluorescent oxygen sensors. In initial efforts, the fluorescent oxygen indicator (pyrene/perylene donor/acceptor pair) was doped into SR membrane with dispersed DMHD–N₂O₂ particles or DACA-6-modified SR membrane.⁹⁸ However, the oxygen response became non-linear because of heterogeneity of the particle-doped film and/or a fluorescence quenching effect from the amine groups of the NO donor. This problem was solved by a two-layer configuration on a quartz slide, which consisted of an underlying DACA-6 modified SR membrane for NO release and an outer SR membrane doped with the oxygen indicator for fluorescence-based oxygen sensing. By using this configuration, only NO gas rather than the NO donor interacted with the oxygen indicator and the linearity and sensitivity toward oxygen was not perturbed. NO release at a flux of >4 × 10⁻¹⁰ mol·cm⁻²·min⁻¹ was obtained for at least 24 h at room temperature and the sensor response was quite stable during this period.

Obviously, the quartz slide-based sensor is not ideal for *in vivo* use. Rather, optical fiber sensor technology has been used successfully for *in vivo* diagnostic applications. Therefore, the two-layer configuration was expanded to an optical fiber type sensor configuration. Dobmeier et al. developed a NO-releasing fiber optical pH sensor.⁹⁹ An aminoalkoxysilane-based xerogel membrane with covalently bonded N-diazeniumdiolates as the NO donor was deposited on the tip of optical fibers. Then the same type of xerogel doped with fluorescent pH indicator, seminaphthorhodamine-1 carboxylate, was coated on the top of the NO donor layer. NO release at $>0.24 \times 10^{-10}$ mol·cm⁻²·min⁻¹ flux was obtained for up to 16 h, indicating the good permeability of NO through the pH-sensitive coating. The underlying NO-releasing layer allows transmission

of excitation and emission light and does not interfere with the fluorescent pH detection. Response to physiologically relevant pH values (7.0-7.8) were obtained by this sensor with a short response time, good linearity, and high precision. Also, *in vitro* platelet adhesion studies confirmed the effectiveness of NO released from such sensor device to inhibit of adhesion of platelets from porcine serum samples.

1.3.3.2 Nitrite for NO generation

To avoid the leaching of NO donors and/or byproducts into the bloodstream, and to produce very controllable NO generation over prolonged time, Ren et al. reported improvement in in vivo analytical performance of intravascular analytical sensors by coupling them with a new electrochemical method of NO generation.¹⁰⁰ This method involves electrochemical reduction of inorganic nitrite ions catalyzed by a copper(II)-ligand complex, copper(II)-tri(2-pyridylmethyl)amine (Cu(II)TPMA; see structure in Fig. 1.2). A dual lumen catheter-type amperometric PO₂ sensor design was developed (with electrochemical NO generation in the bigger lumen and PO₂-sensing platinum-iridium electrode in smaller lumen; see Fig. 1.4) and these devices were implanted in both veins and arteries of rabbits and pigs for up to 21 h. The NO generating sensors exhibited less clot formation and much more accurate analytical results as compared to controls. The advantage of this sensor design is that the amount NO generation can be easily controlled by different applied potentials of between -0.2 and -0.4 V vs. Ag/AgCl to the working electrode of the NO generating lumen. Further, the NO release can then be toggled "on" and "off" by disconnecting the electrode leads from that lumen to the potentiostat.¹⁰¹ Furthermore, the duration of NO generation can be readily extended by increasing the volume or concentration of nitrite within the reservoir. Overall, the study demonstrated that this method could significantly improve



of Figure 1.4. (a) Schematic dual-lumen catheter-type electrochemical NO generating/releasing PO_2 sensor with cross-section geometry of catheter; (b) representative photo illustrating the degree of clot formation on the surface of the control and the NO releasing sensors implanted in rabbit veins for 7 h; (c) average thrombus coverage percentage on NO releasing sensors vs. control sensors. (n = 5 rabbits, p < 0.05); (d) representative current response for a NO releasing sensor (black) and a control sensor (red) (implanted in pig arteries for 21 h) compared with blood draw in vitro test values (reference method, blue squares). Arrows indicate where fraction of inspired oxygen (FiO)₂ changes from 21% to 100%. Figure adapted from Ref. 100.

the biocompatibility of a wide variety of implantable chemical sensors and could provide a potential tool for longer-term *in vivo* investigations into the interplay between long-term NO release and extended *in vivo* analytical performance.

1.3.3.3 S-Nitrosothiols (RSNOs) as NO donors

S-Nitrosothiols (RSNO) are another class of NO donors that can be utilized to improve the biocompatibility of implantable chemical sensors.¹⁰² They are transporters and/or storage vehicles of NO in the body and can be rapidly decomposed to release NO via a several different pathways,¹⁰³⁻¹⁰⁶ including thermal and photolytic cleavage of the S-N bond of the RSNO to release NO. In addition, Cu⁺ species (formed from reduction of Cu²⁺) can catalyze the reaction of RSNOs to liberate NO. Furthermore, ascorbate-mediated reactions can occur in which ascorbate at low

concentration can reduce trace copper ion impurities in solution (i.e., Cu^{2+}) to generate Cu^+ that can subsequently react with RSNOs to produce NO.

Soto et al. were the first to develop implantable chemical sensors with an RSNO as the NO donor source and reported the first long-term investigation on the *in vivo* analytical performance of percutaneously implanted NO-releasing amperometric glucose sensors in swine models.⁷⁶ The sensors were implanted for 10 d and analytical results were obtained comparing the NO-releasing sensors with two different NO-releasing rates with the same total amounts of NO released (3.1 μ mol·cm⁻²): rapid NO release (16.0 ± 4.4 h) from N-diazeniumdiolate NO donor (MAP3/NO)¹⁰⁷ and slower NO release (>74.6 \pm 16.6 h) from S-nitrosothiol-modified silica nanoparticles (MPTMS-RSNO). It was shown that the sensors with slower NO-release rate for extended durations exhibited a significantly lower mean absolute relative deviation (MARD) on days 1 and 3 (26.0 \pm 5.1 and 23.9 \pm 8.6%, respectively) versus controls (34.3 \pm 10.9 and 38.8 \pm 10.4%, respectively), and were characterized by shorter sensor lag time (<4.2 min) in response to intravenous glucose tolerance tests when compared to rapid NO-releasing and controls sensors (>5.8 min) at 3, 7, and 10 d. Overall, both rapid and slower NO-release sensors exhibited improved accuracy vs. controls and it was suggested that materials that are capable of releasing a large amount of NO with even longer duration (i.e., several weeks) would be necessary to create the ultimate NO-release strategy for long-term implantable chemical sensing technology. Indeed, Soto et al. further investigated fabrication of an implantable glucose sensor using an optimized HP-93A-100 PU membrane doped with RSNO-modified silica nanoparticles as the outer, glucose diffusion-limiting layer.¹⁰⁸ The resulting sensor yielded a linear glucose calibration between 1 and 21 mM over a 2-week period while incubating in PBS. Further, the sensing device released NO

at >0.8 pmol·cm⁻²·s⁻¹ for up to 6 d with negligible particle leaching (<0.6%). However, *in vivo* analytical performance of this new design has not yet been reported in the literature.

Of many RSNOs available for use in developing NO release implantable sensors, SNAP (*S*-nitroso-*N*-acetyl-penicillamine; see chemical structure shown in Fig. 1.2, above) has been intensively investigated to prevent donor leaching and stabilize release of NO for improvement in the biocompatibility of various implantable medical devices, including intravascular chemical sensors. Cha et al. (Chapter 5 in this thesis) demonstrated the compatibility of NO with implantable enzymatic amperometric glucose sensors based on the osmium (III/II) mediated electrochemistry and showed that amperometric glucose response is preserved in the presence of continuous NO release from the surrounding SNAP-doped silicone tubing (see Fig. 1.5) when tested in both phosphate buffer (pH 7.4) and *in vitro* within heparinized whole porcine blood at 37 °C (see Chapter 5 for more details).¹⁰⁹

1.3.3.4 Endogenous RSNOs for NO generation

One common limitation of various NO release systems is the finite amount of NO donors that can be held within the sensor device. The NO donor will be depleted typically within a time duration of one day to one month and the biocompatibility benefit will no longer exist after the termination of NO release. To overcome this issue, generation of NO at the surface of biomedical implants from endogenous NO donors such as *S*-nitroso-albumin, *S*-nitroso-glutathione, *S*-nitroso-cysteine, and nitrite has been suggested. Duan and Lewis modified biomedical polymers with L-cysteine and found that such surfaces can generate NO from S-nitroso-albumin via a NO transfer and release mechanism.¹¹⁰ However, the decomposition of nitrosated cysteine may convert L-cysteine



Figure 1.5. (a) Designs of implantable NO-releasing osmium-mediated glucose sensor; (b–d) *in vitro* comparison of intermittent glucose values obtained from a commercial glucometer and continuous glucose measurements obtained from a NO-releasing osmium-mediated glucose sensor as well as a control (No NO release) osmium-mediated glucose sensor poised at +200 mV vs. Ag/AgCl in heparinized whole porcine blood for 3 days. On each day, glucose values were back-calculated using calibration data obtained prior to the *in vitro* blood testing (adapted from 109).

to its disulfide dimer product that is unable to generate NO. Other researchers have examined a series of metal ion-based catalysts embedded within or tethered to a polymer coating of biomedical implants for in situ generation of NO from endogenous RSNOs.¹¹¹⁻¹¹⁷ For example, copper complexes, organoselenium species, organoditelluride species, and copper particles on the implants have been found to catalyze decomposition of various endogenous NO donors in the presence of reductants such as ascorbic acid or glutathione.

Wu et al. adapted the concept of NO generation to intravascular chemical sensors. In this work, 3 μ m and 80 nm Cu⁰ particles were doped into a polyurethane or polyurethane/silicone rubber layer coated on a silicone rubber catheter type amperometric oxygen sensor.¹¹⁷ Slow corrosion of the

 Cu^0 particles generates copper ions, which catalyzes decomposition of GSNO (a naturally present RSNO in blood) in the presence of GSH (also present in blood). The NO-generating oxygen sensors were implanted into porcine arteries for 19– 20 h. A lower degree of thrombus formation and an enhanced oxygen sensing accuracy were observed on such sensors compared to control sensors fabricated without the Cu^0 particles coating. However, it was found that the effectiveness of this NO generation method can vary between animal species/subjects. Some animals show significant clots on the Cu^0 particle-modified catheters, suggesting that these animals may have a lower amount of circulating endogenous RSNOs (probably GSNO) in blood stream and are unable to generate high enough NO levels at the surface of implantable sensors to prevent thrombus formation.

1.4 Summary

The prospect of developing NO-releasing or generating real-time continuous monitoring sensors that respond to clinically important ions, gases and molecules in blood or subcutaneous fluid is an ambitious effort to solve a very challenging analytical problem. A variety of NO release/generation strategies have been shown to release or generate sufficient amounts of NO to enhance the biocompatibility and *in vivo* analytical performance of both intravascular and subcutaneous chemical sensors. However, additional efforts are still needed to devise a platform that can achieve stable release or sufficient generation NO levels for 1-2 weeks and also be compatible with the several different types of sensing chemistries and sensor designs required for continuously monitoring across the spectrum of the clinically important blood analytes. If such sensors can ultimately be realized through continued research efforts, then commercial

development of *in vivo* for routine *in vivo* monitoring would benefit patients in intensive care units around the world.

1.5 Statement of Research

As previously described in the chapter, many efforts have contributed to the development of noninvasive and continuous glucose monitoring systems. Recent studies of non-invasive glucose monitoring systems utilizing glucose in tear fluids have been limited by their ability to achieve the low micromolar range LOQ and high selectivity required using a microliter volume of tear fluid sample. Therefore, the first goal of this thesis is to employ a new method involving blood glucose test strips, to investigate whether any of these strips can achieve the LOQ values required for tear glucose measurements. As reported in Chapter 2, the Roche Accu-Chek type test strips were found to have very low LOQ values and can potentially be applied for human studies.

The second part of this dissertation will focus on investigating the compatibility of nitric oxide release with implantable/continuous osmium(III/II)-mediated glucose sensors. Since SNAP was found to be exceptionally stable when doped into low water uptake biomedical polymers, such as silicone, SNAP is chosen as the nitric oxide source to be coupled with the osmium-mediated glucose sensor for potential application as intravascular/continuous glucose sensors. This dissertation includes *in vitro and in vivo* studies using such sensors to further investigate the validity of the design of NO-releasing osmium-mediated glucose sensor.

Overall, this chapter (Chapter 1) has summarized and reviewed, in detail, the research efforts to date regarding the development of non-invasive and implantable/continuous glucose monitoring

system with an emphasis on the use of NO releasing polymer materials to enhance biocompatibility of implantable glucose sensors. A major portion of this chapter was published in *Applied Materials Today* (K. H. Cha, X. Wang, M. E. Meyerhoff, *Appl. Mater. Today* **2017**, *9*, 589-597).

Chapter 2 reports on evaluation of a number of commercial glucometer test strips for potential measurement of glucose in tears and their potential application in clinical studies involving human studies. This work was published in *Analytical Chemistry* (K. H. Cha, G. C. Jensen, A. S. Balijepalli, B. E. Cohan, M. E. Meyerhoff, *Anal. Chem.* **2014**, *83(3)*, 1902-1908).

Chapter 3 focuses on an investigation into the origin of the very low detection limit and high selectivity toward glucose exhibited by Roche Accu-Chek test strips that enables measurement of tear glucose levels. The conclusions of this study suggest that the unique combination of a nitrosoaniline derivative and a recombinant PQQ-dependent glucose dehydrogenase are the key elements that make the Roche strip so selective and sensitive, enabling direct tear glucose measurements with such strips. This work was published in *Electroanalysis* (K. H. Cha, Y. Qin, M. E. Meyerhoff, *Electroanalysis* **2015**, *27(3)*, 670-676).

Chapter 4 provides data from a further investigation into the potential correlation between tear and blood glucose levels in a small clinical study involving diabetic human subjects and reports a moderate correlation between tear and blood glucose in most of the patients tested. However, the study results also suggest that a longer-term analysis of individual patients (beyond the 1.5 h test period used in this study) may be key to definitively establish the correlation between tear and blood glucose levels.

Chapter 5 focuses on studying compatibility of nitric oxide release from SNAP with implantable enzymatic sensors based on osmium (III/II) mediated electrochemistry and reports that such sensors maintain their functionality, sensitivity and accuracy for detecting glucose levels in blood in the presence of NO release that is high enough to prevent clotting for at least 3 days. This work was published in *ACS Sensors* (K. H. Cha, M. E. Meyerhoff, *ACS Sensors* 2017, *2(9)*, 1262-1266).

Chapter 6 summarizes the major findings/conclusions from this dissertation research and provides suggestions for future directions that will build upon the studies reported herein.

1.6 References

- (1) Danaei, G.; Finucane, M. M.; Lu, Y.; Singh, G. M.; Cowan, M.J.; Paciorek, C.J. *Lancet* **2011**, *378*, 31.
- (2) Hones, J.; Muller, P.; Surridge, N. Diabetes Technol. Ther. 2008, 10, S10.
- (3) Minder, A. E.; Albrecht, D.; Schäfer, J.; Zulewski, H. *Diabetes Res. Clin. Pr.* **2013**, *101*, 57.
- (4) Lane, J. D.; Krumholz, D. M.; Sack, R. A.; Morris, C. *Curr. Eye Res.* 2006, *31*, 895.
- (5) K. H. Cha, X. Wang, M. E. Meyerhoff, *Appl. Mater. Today* **2017**, *9*, 589.
- (6) Maruo K.; Oota, T.; Tsurugi, M.; Nakagawa, T.; Arimoto, H.; Hayakawa, M.; Tamura, M.; Ozaki, Y.; Yamada, Y. *Appl. Spectrosc.* **2006**, *60*, 1423.
- (7) Mueller M; Grunze, M.; Leiter, E. H.; Reifsnyder, P. C.; Klueh, U.; Kreutzer, D. Sens. Actuators, B: Chem. 2009, 142, 502.
- (8) Lambert, J. L.; Morookian, J. M.; Sirk, S. J.; Borchert, M. S. J. *Raman Spectrosc.* 2002, *33*, 524.
- (9) Enejder, A. M. K.; Scecina, T. G.; Oh, J.; Hunter, M.; Shih, W. C.; Sasic, S.; Horowitz, G. L.; Feld, M. S. J. Biomed. Opt. 2005, 10, 031114.
- (10) Cho, O. K.; Kim, Y. Y.; Mitsumaki, H.; Kuwa, K. Clin. Chem. 2004, 50, 1894.
- (11) Lee, H.; Song, C.; Hong, Y. S.; Kim M. S.; Cho, H. R.; Kang, T.; Shin, K.; Choi, S. H.; Hyeon, T.; Kim, D. *Sci. Adv.* **2017**, *3*, e1601314.
- (12) do Amaral, C. E. F.; Wolf, B. Med. Eng. Phys. 2008, 30, 541.
- (13) Evans, N. D.; Gnudi, L.; Rolinski, O. J.; Birch, D. J.; Pickup, J. C. *Diabetes Technol. Ther.* **2003**, *5*, 807.
- (14) Katika, K. M.; Pilon, L. Appl. Opt. 2007, 46, 3359.
- (15) Aslan, K.; Lakowicz, J. R.; Szmacinski, H.; Geddes, C. D. J. Fluoresc. 2004, 14, 677.
- (16) Amarie, D.; Alileche, A.; Dragnea, B.; Glazier, J. A. Anal. Chem. 2010, 82, 343.
- (17) Potts R. O.; Tamada, J. A.; Tierney, M. J. Diabetes-Metab. Res. 2002, 18, S49.
- (18) Lewis, J. G. Br. Med. J. 1957, 1, 585.
- (19) Yan, Q.; Peng, B.; Su, G.; Cohan B. E.; Major, T. C.; Meyerhoff, M. E. *Anal. Chem.* **2011**, *83*, 8341.
- (20) Hasslacher, C. J. Diabetes 2012, 4, 291.
- (21) Wang, X. J. Diabetes. Sci. Technol. 2011, 5, 166.
- (22) Baca, J. T.; Finegold, D. N.; Asher, S. A. Ocul. Surf. 2007, 5, 280.
- (23) Jin, Z.; Chen, R.; Colon, L. A. Anal. Chem. 1997, 69, 1326.
- (24) Badugu, R.; Lakowicz, J. R.; Geddes, C. D. Talanta 2005, 65, 762.
- (25) Chen, L.; Tse, W. H.; Chen, Y.; McDonald, M. W.; Melling, J.; Zhang, J. Biosens. Bioelectron. 2017, 91, 393.
- (26) Baca, J. T.; Taormina, C. R.; Feingold, E.; Finegold, D. N.; Grabowski, J. J.; Asher, S. A. *Clin. Chem.* **2007**, *53*, 1370.
- (27) Yang, X. P.; Pan, X. H.; Blyth, J.; Lowe, C. R. Biosens. Bioelectron. 2008, 23, 899.
- (28) Kagie, A.; Bishop, D. K.; Burdick, J.; La Belle, J. T.; Dymond, R.; Felder, R.; Wang, J. *Electroanalysis* 2008, 20, 1610.
- (29) Chu, M. X.; Kudo, H.; Shirai, T.; Miyajima, K.; Saito, H.; Morimoto, N.; Yano, K.; Iwasaki, Y.; Akiyoshi, K.; Mitsubayashi, K. *Biomed. Microdevices* **2009**, *11*, 837.
- (30) Agustini, D.; Bergamini, M. F.; Marcolino-Junior, L. H. *Biosens. Bioelectron.* **2017**, *98*, 161.
- (31) Badugu, R.; Lakowicz, J. R.; Geddes, C. D. J. Fluoresc. 2004, 14, 617.
- (32) Badugu, R.; Lakowicz, J. R.; Geddes, C. D. Curr. Opin. Biotechnol. 2005, 16, 100.
- (33) Ruan, J.; Chen, C.; Shen. J.; Zhao, X.; Qian, S. Zhu, Z. Polymers 2017, 9, 125.
- (34) Wilson, G. S.; Hu, Y. Chem. Rev. 2000, 100, 2693.
- (35) Frost, M.; Meyerhoff, M. E. Curr. Opin. Chem. Biol. 2002, 6, 633.
- (36) Wilson, G. S.; Gifford, R. *Biosens. Bioelectron.* **2005**, *20*, 2399.
- (37) Mahutte, C. K. Clin. Biochem. 1998, 31, 119.
- (38) Wisniewski, N.; Moussy, F.; Reichert, W. M. Fresenius J. Anal. Chem. 2000, 366, 611.
- (39) Frost, M.; Meyerhoff, M. E. Anal. Chem. 2006, 78, 7370.
- (40) Meyerhoff, M. E. Trends Analyt. Chem. 1993, 12, 257.
- (41) Gorbet, M. B.; Sefton, M. V. Biomaterials **2004**, *25*, 5681.
- (42) Brisbois, E. J.; Handa, H.; Meyerhoff, M. E. Springer International Publishing, Cham, 2015, 481.
- (43) Wu, Y.; Meyerhoff, M. E. *Talanta* **2008**, *75*, 642.
- (44) Andrade, J. D.; Hlady, V. Adv. Polym. Sci. 1986, 79, 1.
- (45) Wisniewski, N.; Reichert, M. Colloids Surf. B: Biointerf. 2000, 18, 197.
- (46) Gerritsen, M.; Jansen, J. A.; Lutterman, J. A. Neth. J. Med. 1999, 54, 167.
- (47) Anderson, J. M. Cardiovasc. Pathol. 1993, 2, 33.
- (48) Sharkawy, A. A.; Klitzman, B.; Truskey, G. A.; Reichert, W. M. *Biomed. Mater. Res.* **1997**, *37*, 401.
- (49) Dungel, P.; Long, N.; Yu, B.; Moussy, Y.; Moussy, F. J. Biomed. Mater. Res. 2008, 85A, 699.
- (50) Klueh, U.; Dorsky, D. I.; Kreutzer, D. L. Biomaterials 2005, 26, 1155.
- (51) Hetrick, E. M.; Prichard, H. L.; Klitzman, B.; Schoenfisch, M. H. *Biomaterials* **2007**, *28*, 4571.
- (52) Nichols, S. P.; Koh, A.; Brown, N. L.; Rose, M. B.; Sun, B.; Slomberg, D. L.; Riccio, D.

A.; Klitzman, B.; Schoenfisch, M. H. Biomaterials 2012, 33, 6305.

- (53) Soto, R. J.; Schoenfisch, M. H. J. Diabetes Sci. Technol. 2015, 9(5), 978.
- (54) Kvist, P. H.; Jensen, H. E. J. Diabetes Sci. Technol. 2007, 1(5), 746.
- (55) Lin, P.; Lin, C.; Mansour, R.; Gu, F. Biosens. Bioelectron. 2013, 47, 451.
- (56) Teo, A. J. T.; Mishra, A.; Park, I.; Kim, Y.; Park, W.; Yoon, Y. ACS Biomater. Sci. Eng. 2016, 2 (4), 454.
- (57) Wang, X. H. R. Lazinica (Ed.), InTech, Rijeka, Croatia, 2013, pp. 111.
- (58) Koh, A.; Nichols, S. P.; Schoenfisch, M. H. J. Diabstes Sci. Technol. 2011, 5, 1052.
- (59) Nichols, S. P.; Koh, A.; Storm, W. L.; Shin, J. H.; Schoenfisch, M. H. Chem. Rev. 2013. 113, 2528.
- (60) Kolobow, T.; Stool, E. W.; Weathersby, P. K.; Pierce, J.; Hayano, F.; Suaudeau, J. Trans. Am. Soc. Artif. Int. Organs 1974, 20A, 269.
- (61) Szycher, M. J. Biomater. Appl. 1988, 3, 297.
- (62) Edmunds Jr., L. H. ASAIO J. 1995, 41, 824.
- (63) Didisheim, P. ASAIO J. **1994**, 40(2), 230.
- (64) Kim, W. S.; Jacobs, H. Blood Purif. 2004, 14(5), 357.
- (65) Larm, O.; Larsson, R.; Olsson, P. Biomater. Med. Devices Artif. Organs 1983, 11(2-3), 161.
- (66) Gutowska, A.; Bae, Y. H.; Jacobs, H.; Mohammad, F.; Mix, D.; Feijen, J.; Kim, S. W. J. *Biomed. Mater. Res.* **1995**, *29*, 811.
- (67) Peppas, N. A.; Huang, Y.; Torres-Lugo, M.; Ward, J. H.; Zhang, J. Annu. Rev. Biomed. Eng. 2000, 2, 9.
- (68) Ayres, N.; Holt, D. J.; Jones, C. F.; Corum, L. E.; Grainger, D. W. J. Polym. Sci., Part A: Polym. Chem. 2008, 46, 7713.
- (69) Mowery, K. A.; Schoenfisch, M. H.; Saavedra, J. E.; Keefer, L. K.; Meyerhoff, M. E. *Biomaterials* **2000**, *21*, 9.
- (70) Zhou, Z.; Meyerhoff, M. E. *Biomaterials* **2005**, *26*, 6506.
- (71) Wu, B.; Gerlitz, B.; Grinnell, B. W.; Meyerhoff, M. E. Biomaterials 2007, 28, 4047.
- (72) Hetrick, E. M.; Schoenfisch, M. H. Annu. Rev. Anal. Chem. 2009, 2, 409.
- (73) Wang, Y.; Chen, S.; Pan, Y.; Gao, J.; Tang, D.; Kong, D.; Wang, S. J. Mater. Chem. B **2015**, *3*, 9212.
- (74) Taite, L. J.; Yang, P.; Jun, H. W.; West, J. L. J. Biomed. Mater. Res. B: Appl. Biomater. 2008, 84, 108.
- (75) Geenen, I. L. A.; Molin, D. G. M.; van den Akker, N. M. S.; Jeukens, F.; Spronk, H. M.; Schurink, G. W. H.; Post, M. J. *J. Tissue Eng. Regen. Med.* 2015, *9*, 564.
- (76) Carpenter, A. W.; Schoenfisch, M. H. Chem. Soc. Rev. 2012, 41, 3742.
- (77) Soto, R. J.; Privett, B. J.; Schoenfisch, M. H. Anal. Chem. 2014, 86, 7141.
- (78) Vaughn, M.; Kuo, L.; Liao, J. J. Cell. Physiol. 1998, 274(6), H2163.
- (79) Hunter, R. A.; Storm, W. L.; Coneski, P. N.; Schoenfisch, M. H. Anal. Chem. 2013, 85(3), 1957.
- (80) Ren, H.; Coughlin, M. A.; Major, T. C.; Aiello, S.; Pena, A. R.; Bartlett, R. H.; Meyerhoff, M. E. Anal. Chem. 2015, 87(16), 8067.
- (81) Nichols, S. P.; Le, N. N.; Klitzman, B.; Schoenfisch, M. H. Anal. Chem. 2011, 83, 1180.
- (82) Drago, R. S.; Karstetter, B. R. J. Am. Chem. Soc. 1961, 83, 1819.
- (83) Hrabie, J. A.; Keefer, L. K. Chem. Rev. 2002, 102, 1135.
- (84) Frost, M. C.; Reynolds, M. M.; Meyerhoff, M. E. Biomaterials 2005, 26, 1685.

- (85) Espadas-Torre, C.; Oklejas, V.; Mowery, K.; Meyerhoff, M. E. J. Am. Chem. Soc. 1997, 119, 2321.
- (86) Mowery, K. A.; Schoenfisch, M. H.; Baliga, N.; Wahr, J. A.; Meyerhoff, M. E. *Electroanalysis* 1999, 11, 681.
- (87) Schoenfisch, M. H.; Mowery, K. A.; Rader, M. V.; Baliga, N.; Wahr, J. A.; Meyerhoff, M. E. Anal. Chem. 2000, 72, 1119.
- (88) Frost, M. C.; Rudich, S. M.; Zhang, H. P.; Maraschio, M. A.; Meyerhoff, M. E. Anal. Chem. 2002, 74, 5942.
- (89) Meruva, R. K.; Meyerhoff, M. E. Biosens. Bioelectron. 1998, 13, 201.
- (90) Frost, M. C.; Batchelor, M. M.; Lee, Y. M.; Zhang, H. P.; Kang, Y. J.; Oh, B. K.; Wilson, G. S.; Gifford, R.; Rudich, S. M.; Meyerhoff, M. E. *Microchem. J.* 2003, 74, 277.
- (91) Gifford, R.; Batchelor, M. M.; Lee, Y.; Gokulrangan, G.; Meyerhoff, M. E.; Wilson, G. S. *J. Biomed. Mater. Res. A* **2005**, *75*, 755.
- (92) Yan, Q.; Major, T. C.; Bartlett, R. H.; Meyerhoff, M. E. *Biosens. Bioelectron.* 2011, 26, 4276.
- (93) Wolf, A. K.; Qin, Y.; Major, T. C.; Meyerhoff, M. E. Chin. Chem. Lett. 2015, 26, 464.
- (94) Marxer, S. M.; Robbins, M. E.; Schoenfisch, M. H. Analyst 2004, 130, 206.
- (95) Shin, J. H.; Marxer, S. M.; Schoenfisch, M. H. Anal. Chem. 2004, 76, 4543.
- (96) Oh, B. K.; Robbins, M. E.; Nablo, B. J.; Schoenfisch, M. H. *Biosens. Bioelectron.* 2005, 21, 749.
- (97) Koh, A.; Lu, Y.; Schoenfisch, M. H. Anal. Chem. 2013, 85, 10488.
- (98) Schoenfisch, M. H.; Zhang, H. P.; Frost, M. C.; Meyerhoff, M. E. Anal. Chem. 2002, 74, 5937.
- (99) Dobmeier, K. P.; Charville, G. W.; Schoenfisch, M. H. Anal. Chem. 2006, 78, 7461.
- (100) Ren, H.; Colletta, A.; Koley, D.; Wu, J.; Xi, C.; Major, T. C.; Bartlett, R. H.; Meyerhoff, M. E. *Bioelechem.* 2015, 104, 10.
- (101) Ren, H.; Wu, J.; Xi, C.; Lehnert, N.; Major, T. C.; Bartlett, R. H.; Meyerhoff, M. E. *ACS Appl. Mater. Interfaces* **2014**, *6*, 3779.
- (102) Jen, M. C.; Serrano, M. C.; van Lith, R.; Ameer, G. A. Adv. Funct. Mater. 2012, 22, 239.
- (103) Williams, D. L. H. Chem. Commun. 1996, 10, 1085.
- (104) Williams, D. L. H. Acc. Chem. Res. 1999, 32, 869.
- (105) Szacilowski, K.; Stasicka, Z. Prog. React. Kinet. Mech. 2001, 26, 1.
- (106) Singh, R. J.; Hogg, N.; Joseph, J.; Kalyanaraman, B. J. Biol. Chem. 1996, 271, 18596.
- (107) Koh, A.; Riccio, D. A.; Sun, B.; Carpenter, A. W.; Nichols, S. P.; Schoenfisch, M. H. *Biosens. Bioelectron.* 2011. 28, 17.
- (108) Soto, R. J.; Schofield, J. B.; Walter, S. E.; Malone-Povolny, M. J.; Schoenfisch, M. H. *ACS Sens.* **2017**, *2*, 140.
- (109) Cha, K. H.; Meyerhoff, M. E. ACS Sens. 2017, 2(9), 1262.
- (110) Duan, X.; Lewis, R. S. Biomaterials 2002, 23, 1197.
- (111) Oh, B. K.; Meyerhoff, M. E. J. Am. Chem. Soc. 2003, 125, 9552.
- (112) Oh, B. K.; Meyerhoff, M. E. Biomaterials 2004, 24, 283.
- (113) Hwang, S.; Cha, W.; Meyerhoff, M. E. Angew. Chem. Int. 2006, 45, 2745.
- (114) Hwang, S.; Meyerhoff, M. E. Biomaterials 2008, 29, 2443.
- (115) Hwang, S.; Meyerhoff, M. E. J. Mater. Chem. 2007, 17, 1462.
- (116) Cha, W.; Meyerhoff, M. E. Biomaterials 2007, 28, 19.
- (117) Wu, Y.; Rojas, A. P.; Griffith, G. W.; Skrzypchak, A. M.; Lafayette, N.; Bartlett, R. H.;

Meyerhoff, M. E. Sens. Actuators B: Chem. 2007, 121, 36.

CHAPTER 2

Evaluation of Commercial Glucometer Test Strips for Potential Measurement of Glucose in Tears

Cha, K.H.; Jensen, G.C.; Balijepalli, A.S.; Cohan, B.E.; Meyerhoff, M.E. *Anal. Chem.* 2014; 86 (3): 1902-1908 *

2.1 Introduction

The World Health Organization (WHO) reports that more than 347 million people worldwide live with diabetes. Over 1.1 million deaths in 2005 were caused by diabetes, and deaths attributed to diabetes are estimated to double by 2030.¹ Regular measurement and control of blood glucose is essential to avoid life-threatening hyper- and hypoglycemic events and associated serious long-term complications.² Normally, a personal blood glucometer with a test strip allows diabetics to monitor their blood glucose level by obtaining a small sample of capillary blood with a lancet, either from the fingertip or forearm. Optimal metabolic control is possible for these patients with

^{*} The principal experiments of this chapter were conducted by Kyoung Ha Cha, with the help of Dr. Gary C. Jensen in understanding the design of glucometer test strips, Anant Balijepalli in helping perform the clinical study, and Dr. Bruce E. Cohan in funding the entire study.

a minimum of four checks per day.³ However, for children and adolescents with Type 1 diabetes on insulin therapy, this method of glucose monitoring is often recommended up to eight times a day.⁴ The resulting discomfort can limit patient compliance and lead to suboptimal blood glucose control⁴ as well as the risk of hypoglycemia. The development of a pain-free, non-invasive sensing technique would greatly improve patient compliance and, ultimately, patient outcomes.

Many studies have aimed to establish a useful yet noninvasive technique for blood glucose level monitoring. The methods applied include infrared (IR) spectroscopy,⁵⁶ Raman spectroscopy,⁷⁸ optical coherence tomography, and measurement of tissue metabolic heat conformation;⁹ however, none are currently proven to be clinically applicable.¹⁰ In addition, several *in vivo* sensing approaches have been proposed, including fluorescence spectroscopy,^{11,12} surface plasmon resonance of nanoparticles,^{11,14} electrical impedance measurements, and implantable/subcutaneous amperometric glucose sensors.¹⁵ Among these, only subcutaneous electrochemical glucose sensors are available commercially, although frequent calibration, short *in vivo* lifetimes, limited accuracy, and high cost have prevented their widespread use.¹⁵

Measurement of tear glucose concentrations is an emergent research focus, and several studies have reported a possibly clinically relevant correlation between blood and tear glucose levels.¹⁶⁻¹⁹ Beca et al. recently reviewed studies of this correlation through different detection methods,²⁰ concluding that there is evidence of a correlation; however, further studies are needed to determine the potential utility of tear glucose measurements to help achieve tight, yet safe, glycemic control.

Methods for tear glucose measurement mandate a low micromolar range LOQ, high selectivity over potential interferences (e.g., ascorbic acid and uric acid), and measurement within μ L sample volumes. Tear glucose has been measured by capillary electrophoresis (CE) coupled with laser-induced fluorescence (LIF),²¹ fluorescence sensors,²² liquid chromatography (LC) coupled with electrospray ionization mass spectrometry (ESI-MS),²³ holographic glucose sensors,²⁴ a miniaturized flexible thick-film flow-cell electrochemical detector,²⁵ and a strip-type flexible biosensor.²⁶ Badugu *et al.* also reviewed the feasibility of using disposable contact lenses to monitor tear glucose.²⁷⁻²⁸ A recent study from the Asher laboratory²³ employed a LC-ESI-MS technique to measure glucose concentrations in 1 μ L tear samples obtained from non-diabetic subjects and reported a median value of 28 μ M with a range from 7 - 161 μ M for this group. For a comprehensive review of electrochemical glucose sensors, including applications related to tear glucose, readers may reference Wang et al.²⁹

Electrochemical measurement of blood glucose with enzyme-based glucometer test strips is the standard method for diabetic near-patient and self-patient care and may be applicable to measure tear glucose concentration. Recently, a sensitive electrochemical technique for measurement of tear glucose, utilizing needle-type amperometric glucose sensors inserted into small glass capillary tubes, was applied to tear samples from anesthetized rabbits.¹⁷ This method provided comparable detection limits to the LC-ESI-MS technique with reduced time, effort, and cost investment. Results showed a clear correlation between blood glucose and tear glucose levels, but the ratio varied from between individual animal subjects, perhaps due to differing tear production rates. However, the electrochemical sensors employed in these studies required 3 - 5 µL sample volumes, which is too large in quantity for application in studies involving human subjects.

Herein, we examine the use of commercial electrochemical blood glucose glucometer strips requiring $< 1 \mu L$ to measure glucose in tears. While these strips are designed to measure blood glucose in the range of 3 - 20 mM, testing strips from several manufacturers with a laboratory potentiostat to control the applied potential and measure low currents showed that strips employing glucose dehydrogenase (GDH) as the active enzyme yield output currents that provide reproducible and linear amperometric responses to glucose in the 0 - 100 µM range. Specifically, strips manufactured by Roche Diagnostics which utilize pyrroloquinoline quinone (PQQ)dependent glucose dehydrogenase (GDH) as the active enzyme and a nitrosoaniline derivative as the electron transfer mediator yield the best analytical performance. Further, optimizing the potential applied to the working electrode in these strips results in relatively low interference from both ascorbic and uric acid.³⁰ It will be shown that measurements of glucose with these strips using tear samples from normal human subjects were found to be in the range reported by Asher's group,²³ suggesting that these strips could provide a simple tool for more extensive clinical studies to determine whether tear glucose sufficiently correlates with blood to be useful as a supplementary method to manage diabetic patients.

2.2 Experimental Section

2.2.1 Chemicals and Reagents

D-(+)-glucose, sodium chloride (NaCl), potassium chloride (KCl), sodium phosphate dibasic (Na₂HPO₄), potassium phosphate monobasic (KH₂PO₄), L-ascorbic acid, uric acid, and acetaminophen were purchased from Sigma-Aldrich (St. Louis, MO).

2.2.2 Sensitivity and Selectivity Analysis

To determine the sensitivity and selectivity of commercial glucometer strips, a four-channel Biostat (ESA Biosciences Inc., Chelmsford, MA) measured the amperometric response to glucose solutions (0 - 100 μ M) and to glucose solutions (0 - 100 μ M) containing potential interfering species (easily oxidizable species at applied potentials). For testing of the commercial glucose strips, the reported applied potentials are versus the internal reference/counter electrode within the strips. The measured currents reported are a ten second average value from an indicated starting measurement time-point following sample introduction. A 5 mM stock solution of glucose was made in phosphate saline buffer (PBS) solution containing 137 mM NaCl, 2.7 mM KCl, 10 mM Na₂HPO₄, and 2.0 mM KH₂PO₄, adjusted pH to 7.4 with 1.0 M NaOH. Standard solutions were prepared by dilution of this stock solution using PBS solution for calibrations of glucose at 0, 50, and 100 μ M. More detailed assessment of low-end analytical performance for the human study employed calibration curves for glucose standards at 10, 50 and 100 μ M. Glucose solutions of 25, 50, 75, and 100 µM containing 10 µM acetaminophen, 100 µM ascorbic acid (freshly made each day), and 100 µM uric acid were used to examine the cumulative effect of these potential interference species on the accuracy of glucose measurements. The reported average levels of ascorbic and uric acid in tears are 20 ± 6.2 and $68 \pm 46 \mu M$,^{31,32} respectively. 10 μM acetaminophen was added to the test mixture, under the prediction that this species may be present in tears at a relative dilution ratio similar to blood glucose, if the patient treatment included this drug.

2.2.3 Human Tear Sample Acquisition

Testing of nine fasting, non-diabetic adult volunteers was conducted in the clinical research laboratory of EyeLab Group under the direction of Dr. Bruce Cohan, M.D. (ophthalmologist). A technician viewed the eye, which was not anesthetized, through a slit lamp microscope at $\times 16$ magnification. Separate tear samples were obtained from the lacrimal lake via hand-held 1 µL disposable glass capillary tubes (micro-pipettes "Microcaps" Drummond Scientific Co., Broomall, PA.) from the right and then the left eye of each subject without intermission. Each subsequent set of samples was obtained after ~ 20 min intervals. Tear sample pairs (from each eye) were obtained during an $\sim 3-4$ h session. For some subjects, samples were obtained on multiple days.

2.2.4 Human Tear Sample Glucose Measurement

A micro tear bubble, produced at the end of the capillary tube by application of gentle force on a rubber bulb dispenser attached to the other end of the capillary tubes, was wicked from the tube into a test-strip (ACCU-CHEK Aviva Plus, LOT #s: 491361, 490702, Roche Diagnostics, Indianapolis, IN) attached to a potentiostat for immediate analysis of glucose concentration. For human tear measurements, average current values over a 10 s period were recorded starting at the 5 s time-point after calibrant or sample introduction were used to prepare the calibration curve and quantitate tear values, respectively.

2.3 Results and Discussion

2.3.1 Sensitivity Analysis of Glucometer Strips

Table 2.1 lists the commercially manufactured glucometer strips evaluated in this research along with the enzyme and mediator employed in each device, the LOQ (μ M), and the R² values obtained

from preliminary studies. The working and reference/counter electrodes for each strip were identified from the manufacturers' patent descriptions and/or the manuals supplied with the strip and/or meters. Strips from each manufacturer were tested with the standard glucose solution for 6 min at an applied voltage of +400 mV vs. Ag/AgCl (n = 5) at physiological temperature (37 °C). For these preliminary studies, a calibration graph (0 - 100 μ M) with the greatest degree of linearity was used to obtain the LOQ by recording the ten second average of current starting at the 60 s time-point after solution was wicked into the strips.

Figures 2.1 show representative current vs. concentration data for several of the commercial strips tested. Standard deviations of the blank and the two glucose standards (50 and 100 μ M), are large with high LOQ values (10x standard deviation for the blank) for the strips that employ FAD/NAD-dependent GDH or GOD (glucose oxidase) as the glucose-selective enzyme: Abbott, Bayer, and

Table	2.1.	Enzyme	and	mediator	combinations	in	commercial	glucometer	strips	and
prelimi	inary	assessmer	nt of o	detection c	apability at low	lev	el glucose co	ncentrations ^α		

Company	Product	Enzyme	Mediator	LOQ (µM)	R ²
Abbott	FreeStyle Lite®	FAD-GDH	Osmium	470	0.9999
Abbott	Precision Xtra®	NAD-GDH	Phenanthroline Quinone	2040	0.6493
Bayer	CONTOUR®	FAD-GDH	Potassium Ferricyanide	287	0.9711
Johnson & Johnson	OneTouch® Verio™	FAD-GDH	Potassium Ferricyanide	374	0.9931
Johnson & Johnson	OneTouch® Ultra® Blue	GOD	Potassium Ferricyanide	954	0.3879
Roche Diagnostics	ACCU-CHEK® Aviva Plus	PQQ-GDH	Nitosoaniline Derivative	45	0.9923
Nipro Diagnostics	TRUEtest®	PQQ-GDH	N/A	100	0.9541

^{α} LOQ and R² from calibrations (0 - 100 μ M) of glucose solution at average current recorded for 10 s starting 60 s following sample introduction (37 °C, +400 mV applied between working and counter electrodes; n = 5).



Figure 2.1. Calibration data for 0, 50 and 100 μ M standard glucose solution using 10 s average current recorded starting at the 60 s time point after sample introduction (at 37 °C, applied voltage to working electrode = +400 mV) for several different commercial glucometer strips: (A) FreeStyle Lite® by Abbott; (B) Precision Xtra® by Abbott; (C) CONTOUR® by Bayer; and (D) OneTouch® Ultra® Blue by Johnson & Johnson. Error bars are standard deviations (S.D.) for n = 5 measurements.

Johnson & Johnson strips (see Table 2.1). Similar results were obtained with other strips which do not utilize PQQ-dependent GDH (data not shown). Thus, all of these strips show inadequate sensitivity at glucose concentrations that mimic tears.

In contrast, strips manufactured with PQQ-GDH, by Roche Diagnostics and Nipro Diagnostics, exhibit far greater precision and linearity in the 0 - 100 μ M glucose range. The lower LOQ values reported in Table 2.1 for these strips indicates better reproducibility of measured currents for the blank solution and standard glucose solutions for the average current recorded over a 10 s period beginning at the 60 s time-point after sample introduction, with the LOQs for these two strips approaching the lower range required for tear measurements.

Initial screening of the glucometer strips indicates that strips utilizing the PQQ-GDH enzyme with gold working and reference/counter electrodes exhibit the best performance at lower-end glucose



Figure 2.2. (A) Dynamic amperometric response (at 37 °C; applied potential = + 400 mV between the working and counter electrodes of the strip; $LOQ = 44 \ \mu\text{M}$) and (B) resulting calibration data for ACCU-CHEK® Aviva Plus strips (Roche Diagnostics) with PBS solution (pH 7.4, 0 μ M glucose, no interference) (black) and glucose concentrations of 50 (red), 100 (blue) μ M) with the 10 s average current recorded 5 s following sample introduction. Calibration data for (C) ACCU-CHEK® Aviva Plus (LOQ = 3 μ M) and (D) TRUEtest® (Nipro Diagnostics) (LOQ = 17 μ M) using PBS solution (0 μ M) and glucose concentrations (50, 100 μ M) measured beginning 5 s following sample introduction (at room temperature (RT); applied potential = +300 mV between the working and counter electrodes). Error bars are S.D. for n = 5 samples.

concentrations, as compared to other strips. Consequently, the Nipro and Roche strips were further tested in more extensive detail for the effects of temperature and interfering species, as well as the influence of applied voltage to the working electrode and the timing of the current measurement after sample application. Figure 2.2A illustrates the current response of Roche strips over the initial 15 s after 0, 50, and 100 μ M glucose standards are wicked into the strips, with an applied potential of +400 mV. The 10 s average current values recorded at 5 s after sample introduction yields calibration data (Figure 2.2B), equivalent to or better (higher current) than data obtained at the average current measured 60 s after sample application. Under these conditions, the LOQ for the Roche strip is 44 μ M (n = 5). Similar results were obtained with Nipro strips (data not shown), with a LOQ of 40 μ M (n = 5).

ACCO-CHERC	9 Aviva	Calculated	[Glucose]	in Inte	erference	Solution	based on
Plus		Calibration	(µM)				
Standard [Gluc	ose] in						
Interference	Solution	+100 mV	125 mV	150 mV	175 mV	200 mV	300 mV
(µM)							
25		28 ± 0	26 ± 0	25 ± 1	24 ± 2	25 ± 1	21 ± 1
50		55 ± 3	51 ± 2	50 ± 1	48 ± 0	51 ± 3	47 ± 2
75		75 ± 2	70 ± 1	69 ± 2	71 ± 2	71 ± 2	69 ± 0
		Calculated	[Glucose]	in Int	erference	Solution	based on
TRUEtest®		Calculated	[Olucose]	III III0		Solution	based off
		Calibration	(µM)				
Standard [Gluc	ose] in	Calibration	(µM)				
Standard [Gluc Interference	ose] in Solution	Calibration 50 mV	<u>(μM)</u> 100 m	V	200 mV	300 r	nV
Standard [Gluc Interference (µM)	ose] in Solution	Calibration 50 mV	(μM) 100 m	V	200 mV	300 r	nV
Standard [Gluc Interference (µM) 25	ose] in Solution	Calibration 50 mV 211 ± 21	(μM) 100 m 236 ±	V 4	200 mV 180 ± 2	300 r 942 ⊧	nV = 140
Standard [Gluc Interference (µM) 25 50	ose] in Solution	Calibration 50 mV 211 ± 21 150 ± 16	(μM) 100 m 236 ± 291 ±	V 4 36	200 mV 180 ± 2 189 ± 8	300 r 942 ± 1050	nV = 140 ± 92
Standard [Gluc Interference (µM) 25 50 100	ose] in Solution	Calibration 50 mV 211 ± 21 150 ± 16 229 ± 14	(μM) 100 m 236 ± 291 ± 361 ±	V 4 36 11	200 mV 180 ± 2 189 ± 8 243 ± 9	300 r 942 ≠ 1050 -190.	nV = 140 ± 92 3

Table 2.2. Selectivity Test Results for ACCU-CHEK® Aviva Plus and TRUEtest® strips^a

^{α}Comparison of selectivity test results for ACCU-CHEK® Aviva Plus and TRUEtest® strips calibrated with standard glucose solutions and tested for measuring standard glucose solution with interfering species present (10 μ M acetaminophen, 100 μ M ascorbic acid, 100 μ M uric acid) at the given applied voltages.

For tests at room temperature (Figure 2.2C and 2.2D), when the potential applied to the working electrode of the strips was changed to +300 mV and current averaged beginning at the 5 s timepoint after sample application, the LOQs for the Roche and Nipro strips were improved to 3 and 17 μ M, respectively (n = 5). Better glucose detection limits were observed for both ACCU-CHEK® Aviva Plus (Roche Diagnostics) and TRUEtest® (Nipro Diagnostics) strips, which may result from the lower K_M value for glucose with PQQ-dependent glucose dehydrogenase³³⁻³⁷ as compared to the other enzymes that catalyze glucose oxidation. The lower K_M results in an increase of the rate of reaction and, therefore, increasing current from oxidation of reduced mediator at the working electrode, at lower glucose concentrations.

2.3.2 Selectivity Analysis for Glucometer Strips



Figure 2.3. Calibration data for 0 - 100 μ M glucose solutions with average currents over 10 s period recorded beginning 5 s after sample introduction (at RT) using (A) ACCU-CHEK® Aviva Plus strips and (B) TRUEtest® strips at different applied voltages. All test solutions contained 10 μ M acetaminophen, 100 μ M ascorbic acid, and 100 μ M uric acid. Both sets of calibration data are stacked in the order of increasing applied potential from bottom to top. Calibrations shown in (A): +100 mV (red, LOQ = 3 μ M, R2 = 0.9949), +125 mV (purple, LOQ = 3 μ M, R2 = 0.9957), +150 mV (black, LOQ = 9 μ M, R2 = 0.9924), 175 mV (orange, LOQ = 19 μ M, R2 = 0.9930), +200 mV (green, LOQ = 18 μ M, R2 = 0.9956), and +300 mV (blue, LOQ = 14 μ M, R2 = 0.9912). Calibrations shown in (B): +50 mV (black, LOQ = 27 μ M, R2 = 0.9989), and +300 mV (blue, LOQ = 64 μ M, R2 = 0.9947). Error bars are S.D. for n = 3 (ACCU-CHEK® Aviva Plus) and n = 5 (TRUEtest®) replicate measurements of each standard solution.

The selectivity of ACCU-CHEK® Aviva Plus and TRUEtest® strips over interferences was tested with glucose solutions in the presence of interfering species that may be found in tears (10 μ M acetaminophen, 100 μ M ascorbic acid, and 100 μ M uric acid) (see Table 2.2). The ACCU-CHEK® Aviva Plus strips with three concentrations of glucose (25, 50, 75 μ M), were tested with interferences at applied voltages in the range of +100 - 300 mV (n = 3) after obtaining calibration currents for glucose only in PBS buffer over this same range. The experiment was also conducted with the TRUEtest® strips with glucose concentrations of 25, 50, 100 μ M and the same interferences but at applied voltages in the range of +50 - 300 mV (n = 5) to find the optimal potential for best selectivity.

Figure 2.3 shows the calibration data for the ACCU-CHEK® Aviva Plus and TRUEtest® strips in the range of 0 - 100 µM, at RT with different applied voltages testing glucose solutions containing



Figure 2.4. Dynamic amperometric response to ACCU-CHEK® Aviva Plus strips with PBS solution (pH 7.4, 0 μ M glucose, no interference) (black) and of glucose at 25 (red), 50 (blue), 75 (green) μ M with interferences present (10 μ M acetaminophen, 100 μ M ascorbic acid, 100 μ M uric acid) for 15 s at +150 mV vs. glucose strip counter electrode. Inset: Corresponding calibration curve toward glucose in the presence of interfering species. Error bars are S.D. for n = 3 replicate measurements of each standard (with interferences) solution.

interferences. Table 2.2 shows data with Roche and Nipro strips calibrated in standard glucose solution and then tested with known glucose solutions containing interfering species at the applied voltages. As shown in Figure 2.3 and Table 2.2, while Nipro strips exhibit good low-end reproducibility and detection limits toward glucose with applied voltages of +50, 100, 200, and 300 mV, large interference occurs at all applied potentials, yielding very large errors in glucose values for samples containing the three interfering species. In contrast, for Roche strips, at all applied potentials, the presence of interfering species yielded relatively small errors in measurement of glucose at levels < 100 μ M.

Figure 2.4 shows the amperometric response to ACCU-CHEK® Aviva Plus strips with PBS solution (pH 7.4, 0 μ M glucose, no interference) and glucose (25, 50, 75 μ M) in interferences (10 μ M acetaminophen, 100 μ M ascorbic acid, 100 μ M uric acid) for 15 s (at +150 mV, n = 3). The

change in current between 25, 50, and 75 μ M glucose in the presence of interfering species is reproducible and displays good linearity (inset).

Figure 2.3 shows the effect of applied voltage for the ACCU-CHEK® Aviva Plus strips on the calibration data over the 1-100 μ M glucose concentration range. Table 2.2 reports errors of \leq 15 % at all applied voltages; however, voltages of 150 mV and 200 mV tend to yield lower percent errors. The LOQ for both +150 mV and +200 mV applied voltages are 9 μ M and 18 μ M, respectively (n = 3). Since triplicate measurements are insufficient to yield true standard deviat ions and, in turn, obtain "true" LOQs, additional measurements were made at +150 mV and +200 mV, which resulted in LOQ values of 16 μ M and 20 μ M, respectively (n = 9). The optimal applied potential for the Roche Strips was found to be +150 mV.

It appears that the Roche ACCU-CHECK strips exhibit dramatically enhanced selectivity versus the Nipro strips which utilize the same PQQ-dependent GDH enzyme largely in part to the specific mediator utilized on the strip. Different mediators exhibit variable reactivity to species such as ascorbic and uric acid, which are easily oxidized. Indeed, the ACCU-CHECK strips utilize a nitrosoanaline derivative, which may be highly resistant to reduction by the ascorbic acid and uric acid species. It is, however, unknown which mediator is employed within the Nipro TRUE-Test strips.

2.3.3 Selectivity Analysis for Different Lots of Glucometer Strips

Preliminary testing revealed lot-to-lot variation in the Roche strip's ability to differentiate glucose from interfering species. For each lot of strips, Roche provides an electronic calibration code

ACCU-CHEK®	Aviva	Calculated	[Glucose]	with Inter	rferences l	based on (Calibration.
Plus		(µM)					
Standard [Glucos Interference (µM)	se] in Solution	Lot# 490702	Lot# 491678	Lot# 491390	Lot# 491345	Lot# 491610	Lot# 491357
25		27 ± 1	22 ± 1	30 ± 3	26 ± 1	29 ± 1	25 ± 1
50		58 ± 2	48 ± 1	52 ± 3	53 ± 2	55 ± 2	50 ± 1
75		81 ± 3	68 ± 1	110 ± 10	75 ± 1	82 ± 3	69 ± 2

Table 2.3. Selectivity test results for six lots of ACCU-CHEK® Aviva Plus strips at +150 mV vs. internal strip counter electrode^a

^{*a*} Comparison of selectivity test results for six different lots of ACCU-CHEK® Aviva Plus strips. Each lot was calibrated with standard pure glucose solutions and tested with the glucose standards containing interferences (10 μ M acetaminophen, 100 μ M ascorbic acid, 100 μ M uric acid) at +150 mV applied voltage (n = 3).

device for insertion into the associated glucometer (for blood measurements), suggesting some differences in the amperometric behavior from lot-to-lot. Therefore, the selectivity of eleven lots of these strips was tested at the optimal applied voltage of +150 mV (n = 3). The calculated errors in the presence of interferences in six of these lots are shown in Table 2.3. Nipro employs a "no-coding" technology in that insertion of the strip itself calibrates the meter. Nipro strips were not tested due to poor selectivity over interferences (see Table 2.2).

Table 2.3 shows errors for all six lots of Roche strips of ≤ 20 % with an applied voltage of +150 mV for all measurements, except the values for lot #491390 at 75 µM glucose. It is not clear why this lot behaved differently at this one higher concentration of glucose. However, differences in interference errors may exist from lot to lot, and thus only certain lots of the Roche strips may yield reliable glucose measurements in tears. Some manufacturing differences can surely exist from lot-to-lot that would give rise to such performance variation. Slight variations in the thickness of the reagent layer, working electrode area, and amount of mediator may not greatly

influence glucose measurements in the range of normal blood glucose concentrations, but could cause more discernible effects in the normal, much lower range of tear glucose concentrations.

2.3.4 Preliminary Assessment of Roche Strips for Measurements of Glucose in Human Tears

Two lots of ACCU-CHEK Aviva Plus strips were determined to possess suitable sensitivity and selectivity to measure glucose in human tear samples. Calibrations of both lots were made in the range of 5 - 100 μ M glucose. LOQs were 7 and 4 μ M, respectively (n = 3). Selectivity tests for these lots exhibited errors of \leq 14 % (n = 3) for 25, 50, and 75 μ M glucose in the presence of 10 μ M acetaminophen, 100 μ M ascorbic acid, and 100 μ M uric acid. Tear samples obtained from n = 9 fasting, non-diabetic human subjects were collected in 1 μ L micro capillary tubes. None of the measured glucose concentrations in human tear samples were found to be below the limit of detection (LOD) of 2 and 1 μ M, respectively (n = 3 for each lot of strips). Table 2.4 lists the average, S.D., median, and range of all measured fasting tear glucose concentrations from each **Table 2.4.** Summary of fasting human tear glucose measurements with ACCU-CHEK Aviva Plus strips for n = 9 non-diabetic volunteers.

Fasting Tear Glucose Concentration (µM)																		
Subject	1	Ą	I	3		С]	D		Е	I	7	(Ĵ	ł	ł	1	ĺ
Eye	R	L	R	L	R	L	R	L	R	L	R	L	R	L	R	L	R	L
Mean	50	52	40	42	36	67	87	77	66	104	23	14	33	32	29	27	22	50
Median	47	51	38	43	38	62	77	76	71	102	23	12	30	32	29	26	22	49
S.D.	10	15	11	12	12	33	34	25	15	30	8	10	20	2	5	1	1	4
n	23	21	24	23	9	9	6	3	5	4	4	3	4	4	4	3	4	3
Patient	30	- 79	16	- 74	22 -	- 139	52 -	- 148	50	- 141	12 -	- 33	14	- 56	25 -	- 35	20-	55
Range																		

^{*la*} Comparison of selectivity test results for six different lots of ACCU-CHEK® Aviva Plus strips. Each lot was calibrated with standard pure glucose solutions and tested with the glucose standards containing interferences (10 μ M acetaminophen, 100 μ M ascorbic acid, 100 μ M uric acid) at 150 mV applied voltage (n = 3).

Table 2.5. Summary of fasting human tear glucose measurements with ACCU-CHEK Aviva Plus strips^{*a*} and LC-ESI-MS from Ref. 23.

	Range (µM)	Mean (µM)	Median (µM)	n (Subject)	n (Measurement)
EyeLab Group	5 - 148	47	43	9	156
Asher	7 - 161	N/A	28	25	148

^aData excludes outliers identified by Grubbs' test.

subject's right and left eyes. Assuming a normal Gaussian distribution for each subject, a total of 7 outliers were identified and excluded from the 163 tear glucose measurements via the Grubbs' outlier test. These outlier values could have resulted from small perturbations induced by the subject rubbing his/her eye in advance and anticipation of tear acquisition or by excess pressure applied to the surface of the eye during the tear acquisition process using the micropipette. In both instances, tiny volumes of the higher glucose levels within tissue or blood capillaries within the eye could enter the tear fluid, temporarily increasing tear glucose values.

Table 2.5 provides the range and median of fasting human tear glucose levels found in this study, and those reported by Asher and coworkers using LC-ESI-MS.²³ With comparable numbers of measurements by the EyeLab and Asher groups, the range of values and the median values found for human tear glucose in non-diabetic subjects for the two data sets are quite comparable, suggesting that the simple glucometer-based method provides tear glucose values that are quite similar to the far more complex LC-MS method.

2.4 Conclusions

Herein, we have demonstrated that test-selected lots of commercially available ACCU-CHEK® Aviva Plus glucometer strips manufactured by Roche exhibit the sensitivity and selectivity necessary to measure low levels of glucose in human tears using sample volumes of $<1 \mu$ L. Preliminary measurements of tear glucose using these strips in non-diabetic volunteers correlate well with those reported by Asher with research laboratory LC-MS instrumentation.²³ The future clinical potential for diabetes management of tear glucose testing with glucometer strips will require thorough evaluation in human subjects including responses to the full range of blood glucose levels during hyperglycemia and, especially, hypoglycemia. The practicality of nearpatient and self-patient tear sample acquisition must also be addressed for safety and efficacy (i.e., incidence of and recovery from perturbation of the sampling site), prior to clinical employment of this measurement method. Certainly, the use of readily-available glucometer test strips in conjunction with a high current sensitivity patient-side measurement system (meter) will greatly aid in carrying out such required research studies.

2.5 References

- (1) Danaei, G.; Finucane, M. M.; Lu, Y.; Singh, G. M.; Cowan, M.J.; Paciorek, C.J. *Lancet* **2011**, *378*, 31.
- (2) Hones, J.; Muller, P.; Surridge, N. *Diabetes Technol. Ther.* **2008**, *10*, S10.
- (3) Minder, A. E.; Albrecht, D.; Schäfer, J.; Zulewski, H. *Diabetes Res. Clin. Pr.* **2013**, *101*, 57.
- (4) Lane, J. D.; Krumholz, D. M.; Sack, R. A.; Morris, C. Curr. Eye Res. 2006, 31, 895.
- (5) Maruo K.; Oota, T.; Tsurugi, M.; Nakagawa, T.; Arimoto, H.; Hayakawa, M.; Tamura, M.; Ozaki, Y.; Yamada, Y. *Appl. Spectrosc.* **2006**, *60*, 1423.
- (6) Mueller M; Grunze, M.; Leiter, E. H.; Reifsnyder, P. C.; Klueh, U.; Kreutzer, D. Sens. Actuators, B: Chem. 2009, 142, 502.
- (7) Lambert, J. L.; Morookian, J. M.; Sirk, S. J.; Borchert, M. S. *J. Raman Spectrosc.* **2002**, *33*, 524.
- (8) Enejder, A. M. K.; Scecina, T. G.; Oh, J.; Hunter, M.; Shih, W. C.; Sasic, S.; Horowitz, G. L.; Feld, M. S. J. Biomed. Opt. 2005, 10, 031114.
- (9) Cho O. K.; Kim, Y. Y.; Mitsumaki, H.; Kuwa, K. Clin. Chem. 2004, 50, 1894.
- (10) do Amaral, C. E. F.; Wolf, B. Med. Eng. Phys. 2008, 30, 541.
- (11) Evans, N. D.; Gnudi, L.; Rolinski, O. J.; Birch, D. J.; Pickup, J. C. *Diabetes Technol. Ther.* **2003**, *5*, 807.
- (12) Katika, K. M.; Pilon, L. Appl. Opt. 2007, 46, 3359.
- (13) Aslan, K.; Lakowicz, J. R.; Szmacinski, H.; Geddes, C. D. J. Fluoresc. 2004, 14, 677.

- (14) Amarie, D.; Alileche, A.; Dragnea, B.; Glazier, J. A. Anal. Chem. 2010, 82, 343.
- (15) Potts R. O.; Tamada, J. A.; Tierney, M. J. Diabetes-Metab. Res. 2002, 18, S49.
- (16) Lewis, J. G. Br. Med. J. 1957, 1, 585.
- (17) Yan, Q.; Peng, B.; Su, G.; Cohan B. E.; Major, T. C.; Meyerhoff, M. E. *Anal. Chem.* **2011**, *83*, 8341.
- (18) Hasslacher, C. J. Diabetes 2012, 4, 291.
- (19) Wang, X. J. Diabetes. Sci. Technol. 2011, 5, 166.
- (20) Baca, J. T.; Finegold, D. N.; Asher, S. A. Ocul. Surf. 2007, 5, 280.
- (21) Jin, Z.; Chen, R.; Colon, L. A. Anal. Chem. 1997, 69, 1326.
- (22) Badugu, R.; Lakowicz, J. R.; Geddes, C. D. Talanta 2005, 65, 762.
- (23) Baca, J. T.; Taormina, C. R.; Feingold, E.; Finegold, D. N.; Grabowski, J. J.; Asher, S. A. *Clin. Chem.* **2007**, *53*, 1370.
- (24) Yang, X. P.; Pan, X. H.; Blyth, J.; Lowe, C. R. Biosens. Bioelectron. 2008, 23, 899.
- (25) Kagie, A.; Bishop, D. K.; Burdick, J.; La Belle, J. T.; Dymond, R.; Felder, R.; Wang, J. *Electroanalysis* 2008, 20, 1610.
- (26) Chu, M. X.; Kudo, H.; Shirai, T.; Miyajima, K.; Saito, H.; Morimoto, N.; Yano, K.; Iwasaki, Y.; Akiyoshi, K.; Mitsubayashi, K. *Biomed. Microdevices* **2009**, *11*, 837.
- (27) Badugu, R.; Lakowicz, J. R.; Geddes, C. D. J. Fluoresc. 2004, 14, 617.
- (28) Badugu, R.; Lakowicz, J. R.; Geddes, C. D. Curr. Opin. Biotechnol. 2005, 16, 100.
- (29) Wang, J. Chem. Rev. 2008, 108, 814.
- (30) Burke, D. W.; Marquant, M.; Zapf, U.; Fritz, M.; Mosoiu, D.; Wilsey, C. Reagent Stripe for Test Strip. U.S. Patent 7,727,467, June 1, **2010**.
- (31) Choy, C. K. M.; Benzie, I. F. F.; Cho, P. Invest. Ophthalmol. Vis. Sci. 2000, 41, 3293.
- (32) Choy, C. K. M.; Cho, P.; Chung, W. Y.; Benzie, I. F. F. Optom. Vis. Sci. 2003, 80, 632.
- (33) Cozier, G. E.; Salleh, R. A.; Anthony, C. *Biochem. J.* **1999**, *647*, 639.
- (34) Yoshida, H.; Kojima, K.; Witarto, AB.; Sode, K. Protein Eng. 1999, 12, 63.
- (35) Kalisz, H. M. Biochem. J. 2000, 349, 553.
- (36) Yuan, X.; Iijima, M.; Oishi, M.; Nagasaki, Y. *Langmuir* **2008**, *24*, 6903.
- (37) Tsujimura, S.; Kojima, S.; Kano, K.; Ikeda, T.; Sato, M.; Sanada, H.; Omura, H. *Biosci. Biotechnol. Biochem.* **2006**, *70*, 654.

CHAPTER 3

Origin of Low Detection Limit and High Selectivity of Roche Accu-Chek Test Strips that Enables Measurement of Tear Glucose Levels

Cha, K.H.; Qin, Yu.; Meyerhoff, M.E. Electroanalysis 2015; 27: 670-676 *

3.1 Introduction

The World Health Organization reports that 347 million people worldwide have diabetes¹ and projects that diabetes will be the 7th leading cause of death in 2030,² despite the advancing modern medical science and technology.^{3,4} Tight glycemic control is essential to prevent or reduce life-threatening hyper- and hypoglycemic crises causing serious long-term complications.⁵ Using a personal blood glucometer with an enzymatically-active disposable test strip, enables diabetic patients to monitor their glucose levels by obtaining a small drop of capillary blood from their fingertips or forearms with a conventional lancet. While minimum of four measurements of glucose per day is recommended to maintain normal levels of glucose in blood,⁶ up to eight checks per day are recommended for

^{*} The key experiments of this chapter were conducted by Kyoung Ha Cha, with help from Dr. Yu Qin in be providing the nitrosoaniline derivative.

children and adolescents with Type 1 diabetes on insulin treatment.⁷ Pricking the finger tip for many times a day can limit patient compliance and develop massive carring/callous formation and loss of sensibility/perception hindrance. Consequently, it may lead to suboptimal blood glucose control and induce hypoglycemia. Therefore, the development and availability of a pain-free glucose measuring device could greatly increase patient compliance and maintain optimal glucose control.

Tear glucose measurements have been suggested as an alternative method to monitor glucose levels. Tear glucose measurement techniques require a low micromolar LOQ, high selectivity over potential interferences such as ascorbic acid, uric acid and acetaminophen, and measurement within microliter or sub-microliter sample volume. Since several studies have shown that there may be a clinically useful correlation between blood and tear glucose levels,⁷⁻⁹ more detailed studies are necessary to confirm the potential utility of tear glucose measurements to help maintain optimal glucose control. Chapter 2 of this dissertation and Cha et al. recently reported¹⁰ that Roche's Accu-Chek blood glucose test strips are capable of glucose level measurements in tear samples with adequate sensitivity (0.127 nA/ μ M) and selectivity with ≤ 13 % error (n = 3) for 25, 50, and 75 µM glucose in the presence of 10 µM acetaminophen, 100 µM ascorbic acid, and 100 µM uric acid. In this previous work, the Roche test strips were used for measurement of glucose concentration in tears from n = 9 normal (non-diabetic) fasting human subjects and reported glucose values were within the range of 5-148 μ M (mean = 47 μ M, median = 43 μ M), a similar range to that reported by Asher and co-workers using an LC-MS technique.⁸ Herein, we utilize bulk phase solution experiments with different enzymes, strip-based screen printed electrode materials and mediators to explain why the Roche test strips exhibit such excellent detection limits

and selectivity compared to all other strips examined in our earlier studies.¹⁰ Results indicate that the unique combination of pyrroloquinoline quinone dependent glucose dehydrogenase (PQQ-GDH) as the active enzyme and a nitrosoaniline derivative as the electron transfer mediator are the key components of this test strip that yield the required sensitivity and selectivity to accurately measure low glucose levels in tear fluid.

3.2 Experimental Section

3.2.1 Chemicals and Reagents

D-(+)-Glucose, glucose oxidase from *A. niger* (149.8 U/mg), sodium chloride (NaCl), potassium ferricyanide(III) (K₃Fe(CN)₆), potassium chloride (KCl), sodium phosphate dibasic (Na₂HPO₄), potassium phosphate monobasic (KH₂PO₄), ascorbic acid, uric acid sodium salt, and acetaminophen were purchased from Sigma-Aldrich (St. Louis, MO). FAD (flavin adenine dinucleotide)-dependent glucose dehydrogenase from Aspergillus sp. (1180 U/mg) was obtained from Sekisui Enzymes (Lexington, MA). PQQ-dependent glucose dehydrogenase (808 U/mg) was a product of Toyobo Enzyme (Osaka, Japan). O-methoxy-[N,N-bis-(2-hydroxyethyl)]-p-nitrosoaniline (nitrosoaniline derivative) (see Fig. 3.1) was generously donated by PharmaBlock R&D Co. Ltd (Nanjing, China).

A four-channel Biostat (ESA Biosciences Inc., Chelmsford, MA) was used to measure the amperometric response to glucose solutions (0-5 mM) containing enzymes, mediators and/or potential interfering species (easily oxidizable species at applied potentials) using an applied potential of +150 mV. For testing electrodes with different materials (palladium, gold and carbon), commercially available glucose test strips from Bayer, Roche Diagnostics and Abbott were



Figure 3.1. Structures of (a) ferricyanide and (b) O-methoxy-[N,N-bis-(2-hydroxyethyl)]-pnitrosoaniline that were examined as electron transfer mediators in this work.

purchased over the counter and the top housing layers were removed to expose the electrochemical sensing sites. Each of these strips use palladium, gold and carbon as the working electrode material, respectively. The exposed surfaces of the strips were then washed thoroughly with deionized water and ethanol to remove any deposited chemical reagents on the surface of electrodes before use in bulk phase solution experiments.

A 1.5 M stock solution of glucose, and 25 mM stock solutions of ascorbic acid, uric acid and acetaminophen were made in phosphate saline buffer (PBS) containing 137 mM NaCl, 2.7 mM KCl, 10 mM Na₂HPO₄, and 2.0 mM KH₂PO₄, adjusted pH to 7.4 with 1.0 M NaOH. Standard solutions were prepared by dilution of the stock solutions using PBS solution for calibrations of glucose. Solutions of 0.5 mM ascorbic and uric acid and 0.2 mM acetaminophen (freshly made each day) were used to examine the cumulative effect of these potential interference species on the accuracy of glucose measurements. The reported average levels of ascorbic and uric acid in tears are 20 ± 6.2 and $68 \pm 46 \mu$ M, respectively.¹¹⁻¹² A 0.2 mM acetaminophen solution was also tested assuming that this species may be present in tears at a relative dilution ratio similar to blood

glucose, if the patient was using this drug. Stock solutions of the various enzymes and mediators were made in PBS solution in the same way.

3.2.2 Electrochemical Measurements

Various glucometer strips for which the reagents layers were removed and the electrodes first exposed and cleaned were used as biamperometric detectors in bulk solution phase experiments. Each of the strips has two identical electrodes that are utilized for glucose detection with one electrode employed as the working electrode (apply positive voltage to oxidize reduced form of mediator formed from enzymatic oxidation of glucose and/or reduction of mediator by potential interferent species that can directly reduce the mediator) and the other serving as the pseudo reference/counter electrode. The cleaned and uncovered electrode ends of the various strips were placed in 5 mL of PBS buffer within a vial and a small magnetic stir bar was utilized to mix the solution. Different mediators (e.g., ferricyanide or the nitrosoaniline derivative (see Fig. 3.1)) alone or in combination with different glucose oxidizing enzymes (at near equivalent activity in term of Units/mL) were added to the PBS solution for testing response to ascorbic acid, uric acid and acetaminophen, as well as to different concentrations of glucose. The Biostat electrochemical analyzer was used to apply the indicated voltage between the two electrodes of the strips and to monitor the resulting current as a function of time after addition of glucose or test interference species.

3.3 Results and Disscusions

3.3.1 Sensor Configuration for Tear Glucose Measurement

A schematic of a generic blood glucometer test strip when used for monitoring glucose in tear fluid is illustrated in Figure 3.2. An assortment of different glucose-oxidizing enzymes have been employed to prepare conventional glucose biosensors for measuring glucose concentration in whole blood. Glucose oxidase (GOX) has been widely used to catalyze oxidation of glucose on or close to the electrode surface, which can be followed by amperometric monitoring of liberated H₂O₂ or the consumed O₂. However, monitoring peroxide production when using GOX requires a high applied potential (+600 mV),¹³ at which potential interferents can also be easily oxidized. Fortunately, via the aid of electron transfer mediators the applied potential required can be significantly lowered to partially reduce the interference problem. Indeed, efficient electron transfer mediators such as ferrocene and its derivatives were originally used for the enzymatic reaction catalyzed by GOX and applied in Abbott's Medisense Precision Q.I.D. blood glucometer test strips back in the early 1990s. However, since then, other mediators including potassium ferricyanide, osmium complexes and nitrosoaniline derivatives have become the mediators of choice and are the ones most widely used in current commercial blood glucometer test strips. In





fact, several commercial strips (e.g., Johnson and Johnson One Touch, Bayer Contour, etc.) utilize potassium ferricyanide as the mediator in the strip configuration illustrated in Figure 3.2. Beyond glucose oxidase, two different types of glucose dehydrogenase (GDH) enzymes are also often employed to create blood glucometer strips; FAD-dependent GDH, and PQQ-dependent GDH. The Roche test strips that were found to be so useful for tear glucose measurements utilize a unique combination of gold working/counter electrodes, PQQ-GDH as an active enzyme and a nitrosoaniline derivative (see Fig. 3.1) as an electron transfer mediator. This strip is able to selectively measure low concentrations glucose found in tears in the presence of electroactive interferents. An optimal applied voltage of +150 mV between the two gold electrodes within the Roche strips for calibration over the 1-100 μ M glucose concentration range was found to yield the lowest % errors from such potential interferent species.¹⁰

3.3.2 Selectivity Analysis

To better understand the enhanced selectivity over potential interferences of the Roche Accu-Chek glucometer strips vs. other common brands when applied for tear measurements, we first examined the response of interfering species (ascorbic acid, uric acid and acetaminophen) on three different electrode strip materials (palladium, gold and carbon) in terms of the steady-state amperometric responses in the presence of two different electron transfer mediators, potassium ferricyanide and the nitrosoaniline derivative (Fig. 3.1), in bulk solution experiments. As stated above, potassium ferricyanide is the most widely used electron transfer mediator at present, and thus it was chosen to be compared directly to the nitrosoaniline derivative. Our aim was to determine the ideal combination yielding the least current from oxidation of the potential interferents. An applied potential of +150 mV between the two strip electrodes was chosen because it is the optimal



Figure 3.3. Comparison of dynamic amperometric responses (at room temperature; applied potential = +150 mV) from the interferents (0.5 mM ascorbic acid (AA), 0.5 mM uric acid (UA) and 0.2 mM acetaminophen (AM)) in the presence of 10 mM potassium ferricyanide (a) or 10 mM nitrosoaniline derivative (b) in PBS (pH 7.4, 0 μ M glucose) on gold-printed electrodes. Background response from 10 mM potassium ferricyanide (a) and nitrosoaniline derivative in PBS solution is recorded between 20-80 s. Response from interferents are recorded between 80-140 s.

potential found previously to achieve the highest selectivity for glucose for the Roche test strips with nitrosoaniline derivative as an electron transfer mediator.¹⁰ Further, this applied potential is close to the +200 mV value that is generally used by the commercial glucometer devices in their intended operation protocols for test strips employing potassium ferricyanide as an electron transfer mediator.¹⁴

As shown in Figure 3.3, a large increase in current is observed on gold-printed electrodes when 0.5 mM ascorbic acid is added at the 80 s mark to the stirred PBS bulk solution (pH 7.4) for sensor strips with either potassium ferricyanide or nitrosoaniline derivative as the redox mediators (+150



Figure 3.4. Comparison of dynamic amperometric responses (at room temperature; applied potential = +150 mV) from the interferents (0.5 mM ascorbic acid (blue), 0.5 mM uric acid (red) and 0.2 mM acetaminophen (green)) in the presence of 10 mM potassium ferricyanide (a, c) or 10 mM nitrosoaniline derivative (b, d) in PBS (pH 7.4, 0 μ M glucose) on palladium (a, b) and carbon electrode (c, d). Background response from 10 mM potassium ferricyanide (a) and nitrosoaniline derivative in PBS solution is recorded between 0-20 s for a, b, and d and 0-80 for c. Response from interferents are recorded afterwards.

mV applied between working and reference/counter electrode of the strips). It should also be noted that the level of currents observed from the addition of ascorbic acid is much greater when ferricyanide is selected at the mediator vs. when an equivalent concentration of nitrosoaniline derivative is employed in the bulk solution (10 mM) under the exact same experimental conditions. However, when the nitrosoaniline derivative is employed (Fig. 3.3b), 0.5 mM ascorbic acid is the only potential interferent species yielding a significant current at +150 mV applied potential (primarily due to reduction of the mediator by ascorbate). As shown in Figure 3.3b, 0.5 mM uric acid and 0.2 mM acetaminophen are not electro-oxidized at all with 10 mM nitrosoaniline derivative in PBS solution. In contrast, when potassium ferricyanide is selected as the mediator, relatively large responses from uric acid and acetaminophen are observed at an applied potential

of +150 mV vs. the counter electrode (Fig. 3.3a). This behavior is also observed for palladium and carbon-printed electrodes (Fig. 3.4). When using potassium ferricyanide as mediator, all three types of electrodes (palladium, gold and carbon) exhibit a significant current increase in the presence of potential interferents, while 0.5 mM uric acid and 0.2 mM acetaminophen do not yield an increase in anodic current with a nitrosoaniline derivative mediator. It is well documented that ascorbic acid can react with the mediators in their oxidized form to form reduced mediator which can then be oxidized electrochemically. Our results indicate that the nitrosoaniline derivative is much more inert to these interfering species than potassium ferricyanide, making the nitrosoaniline derivative a better choice for electron transfer mediator to yield more selective glucose measurements in blood as well as in tear fluid. Initial screening of all possible combinations of different electrode materials and mediators indicates that the use of gold working/counter electrodes along with the nitrosoaniline derivative is the only combination that exhibits the optimal selectivity over potential interferents in the bulk phase experiments with the commerciallybranded, printed electrodes test strips. Consequently, this combination was utilized in further studies to measure low-end glucose concentrations using different glucose oxidizing enzymes as catalysts for detection of glucose.

3.3.3 Sensitivity Analysis

To this end, three different enzymes used for glucose measurements were examined in the bulk phase experiments using biamperometric detection with various glucometer strip electrodes. The enzymes tested included: glucose oxidase and glucose dehydrogenases in combination with two different cofactors/prosthetic groups, flavin adenine dinucleotide (FAD) and pyrroloquinoline quinone (PQQ). The enzymes were added to the PBS buffer alone with either potassium ferricyanide or nitrosoaniline derivative, and the biamperometric response toward additions of glucose concentrations in the range of 0-5 mM were recorded. The response curves toward glucose were obtained initially in PBS without any interfering species present. As shown in Figure 3.5a, with 125 U/mL FAD- or PQQ-dependent GDHs, coupled with 20 mM potassium ferricyanide in the test solution and gold electrodes as the biamperometric detectors (at an applied potential of +150 mV), significant amperometric signals are observed for low-end glucose concentrations in the range of 0-0.2 mM. In contrast, when using GOX at the same U/mL concentrations, there is very poor sensitivity to changes in glucose concentrations, with amperometric response observed only in the higher concentrations range (≥ 0.5 mM). In the case of the nitrosoaniline derivative (Fig. 3.5b), only the PQQ-GDH catalyst yields significant amperometric response to glucose in the low concentration range required for tear glucose measurements. It is important to note that 20 mM of both mediators was used for these experiments, instead of 10 mM, in order to have a sufficient level of mediator necessary to oxidize high-end glucose concentrations.

The far superior glucose sensitivity observed in the above experiments when using both FAD- and PQQ-dependent GDH may relate to their lower Michaelis-Menten constant K_M values, compared to the other enzymes that catalyze glucose oxidation. The lower K_M results in an increased rate of reaction at a lower glucose level and, therefore, increased catalytic current from oxidation of the reduced mediator at the working electrode for lower glucose concentrations. Indeed the Michaelis-Menten constant K_M values toward glucose of GOX, FAD-GDH, NAD-GDH, and PQQ-GDH are reported to be 27, 0.46, 10 and 0.38 mM, respectively.¹⁵⁻¹⁸ Hence, there is a significantly lower K_M value for the two enzymes that exhibit the most sensitive response to glucose levels in the bulk solution phase experiments. Further, from the patent literature,¹⁹ it is known that Roche utilizes a



Figure 3.5. Dynamic amperometric response (at room temperature; applied potential = +150 mV) for changes in glucose concentration (0-5 mM) in the presence of (a) 20 mM potassium ferricyanide and (b) 20 mM nitrosoaniline derivative in PBS solution (pH 7.4) with 125 U/ml glucose oxidase (GOX), FAD- (FAD-GDH), and PQQ-dependent glucose dehydrogenase (PQQ-GDH). The activities of the enzymes were 149.8 U/mg, 1180 U/mg and 808 U/mg, respectively. Insets: Current response from low-end glucose concentrations (0-0.2 mM). Within the figure, the indicated concentrations represent the total concentration of glucose at each arrow point in time.

recombinant form of PQQ-GDH in order to avoid interference from maltose that can cause errors with blood glucometer devices that utilize PQQ-GDH as the active enzyme. Hence, it is possible that the recombinant form of PQQ-GDH utilized in the Accu-Chek strips may have an even lower $K_{\rm M}$ compared to the 0.38 mM value reported for the native enzyme.

Although use of FAD- and PQQ-dependent GDHs in combination with potassium ferricyanide as the electron transfer mediator exhibits very good sensitivity for low concentration detection of glucose, the use of potassium ferricyanide is undesirable due to the considerable amount of interference current produced from ascorbic acid, uric acid, and/or acetaminophen oxidation of, and from this mediator (see Fig. 3.3a, above).

As stated above, when using nitrosoaniline derivative as the electron transfer mediator (Fig. 3.5b), PQQ-GDH produces the most sensitive response to glucose among all three enzymes tested, especially in the desired tear glucose measuring range (0-0.2 mM). With the nitrosoaniline derivative, GOX and FAD-GDH only yield amperometric response to glucose in the higher range of glucose levels (\geq 0.5 mM). It should be noted that for same enzyme, the sensitivity differs significantly between the different mediators. Our results indicate that for FAD-GDH, potassium ferricyanide is a better mediator than nitrosoaniline for facilitating the electron-transfer process. However, for PQQ-GDH, the use of the nitrosoaniline derivative is required to obtain enhanced sensitivity, and the selectivity over interfering species is far superior to the combination of FAD-GDH and potassium ferricyanide.

The rapid decrease in current after every glucose measurement for the combination of 20 mM nitrosoaniline derivative in PBS solution (pH 7.4) with 125 U/ml GOX and PQQ-GDH (as shown in Fig. 3.2b) that exhibit far more rapid response time, may be a result of adsorption of nitrosoaniline derivative onto electrode surface. When bare electrodes were used to measure current in 10 mM potassium ferricyanide/10 mM potassium ferrocyanide solution with an applied potential of +150 mV, steady currents are measured after ~5 minutes (solid lines in Fig. 3.6). H



Figure 3.6. Comparison of dynamic amperometric responses (at room temperature; applied potential = +150 mV) in the presence of 10 mM potassium ferricyanide and 10 mM potassium ferrocyanide before (solid lines) and after (dashed lines) measuring 20 mM nitrosoaniline derivative in PBS solution for 2 h.

owever, when these same electrodes are used to measure in 10 mM potassium ferricyanide/10 mM potassium ferrocyanide solution after measuring the 20 mM nitrosoaniline derivative, higher and unstable currents are obtained (dashed lines in Fig. 3.6) after 2 h with the same applied potential. This suggests that there may be an additional reaction between in 10 mM potassium ferricyanide/10 mM potassium ferrocyanide and the nitrosoaniline derivative that might have been absorbed onto electrode surface. Indeed, the reaction between 10 mM potassium ferricyanide/10 mM potassium ferrocyanide and the nitrosoaniline derivative with an applied potential of +150 mV yields currents in the range of 2-3 μ A, but the 20 mM nitrosoaniline derivative itself yields only ~6 nA with the same applied potential (data not shown).

3.3.4 Optimization for Measurement of Low-end Glucose Level

The electrochemical response of the strips made with gold electrodes to glucose in the presence of ascorbic acid in the bulk solution was also examined using the optimal combination of PQQ-GDH


Figure 3.7. Dynamic amperometric response (at room temperature; applied potential = +150 mV) for increasing glucose concentrations (0-1 mM) in the presence of 10 and 3.7 mM nitrosoaniline derivative (ND), PQQ-dependent glucose dehydrogenase with various concentration of ascorbic acid (AA; 0, 0.1 and 0.5 mM) in PBS solution (pH 7.4). Within the figure, concentrations refer to the total concentration of glucose made at each arrow point in time. (b) Dependence of steady-state current on glucose concentration in the presence of 0.1 mM ascorbic acid (circles) and without ascorbic acid (squares).

and the nitrosoaniline derivative. The previous results (see Fig. 3.3) suggested that ascorbic acid still poses a problem and that the combination of PQQ-GDH, nitrosoaniline derivative, and gold electrode detection cannot avoid oxidation of ascorbic acid at an applied potential of +150 mV. As shown in Figure 3.7a, with 10 mM nitrosoaniline derivative present in solution, the glucose sensitivity at low concentrations is better in the presence of 0.1 mM ascorbic acid than 0.5 mM ascorbic acid. However, as also shown in Figure 3.7a, the interference effect from ascorbic acid can be partially minimized by controlling the concentration of the mediator, through which the reaction between a lower concentration of nitrosoaniline derivative (3.7 mM) and 0.1 mM ascorbic acid yields less anodic current than the reaction between 10 mM nitrosoaniline derivative and 0.1 mM ascorbic acid. Figure 3.7b shows the resulting glucose calibration curves for the gold-printed electrodes coupled with PQQ-GDH and 3.7 mM nitrosoaniline derivative (in bulk solution) in the absence and presence of 0.1 mM ascorbic acid. Although in the evaluation range of 0-50 μ M glucose, the oxidation current from ascorbic acid is high, at the 50 μ M glucose level, the current increase by the presence of 0.1 mM ascorbic acid is ca. 27 %, corresponding to +13 μ M error in

the glucose measurement. At 200 μ M glucose, the error from ascorbic acid is only 1%. However, in prior work with fully assembled Roche Accu-Chek strips,¹⁰ much lower errors were observed when ascorbate at 0.1 mM was combined with acetaminophen and uric acid as an interference solution in the presence of 50 μ M glucose (combined error from interferences <13%). Since in the fully assembled commercial blood glucometer strip, the PQQ-GDH and nitrosoaniline derivative (1:2 by weight) are coated over the surface of the gold electrodes within a polymeric matrix containing thickeners stabilizers, detergents, and other adjunct materials, and the current is recorded after 5 s after wicking of the test samples into the strips, there may well be some partial mass transport rejection of ascorbic acid into the enzyme/mediator layer that further enhances selectivity compared to the homogeneous experiments reported in this work.

3.4 Conclusions

We have demonstrated that that the ability of Roche Accu-Chek test strips to be used to measure μ M levels of glucose without substantial interference from other redox active species results from the unique combination of using PQQ-GDH as the active enzyme in combination with nitrosoaniline derivative as an electron transfer mediator. The low K_M value of this enzyme, coupled with the inability of uric acid and acetaminophen to reduce the nitrosoaniline mediator and yield electrochemical response in a biamperometric measurement mode, appear to be the key elements that provide the excellent LOQ value and selectivity of these strips. These findings suggest that any future efforts to fabricate specialized electrochemical glucometer type test strips, specifically for direct sampling of sub-microliter quantities of tear fluid from eyes, should utilize PQQ-GDH in combination with O-methoxy-[N,N-bis-(2-hydroxyethyl)]-p-nitrosoaniline as the reagents.

3.5 References

- Danaei, G.; Finucane, M. M.; Lu, Y.; Singh, G. M.; Cowan, M. J.; Paciorek, C. J.; Lin, J. K.; Farzadfar, F.; Khang, Y. H.; Stevens, G. A.; Rao, M.; Ali, M. K.; Riley, L. M.; Robinson, C. A.; Ezzati, M. *Lancet* 2011, *378*, 31.
- (2) Global status report on noncommunicable diseases 2010. Geneva, World Health Organization, 2011.
- (3) Zimmet, P.; Alberti, K. G. M. M.; Shaw, J. *Nature* 2001, *414*, 782.
- (4) Brownlee, M. *Nature* **2001**, *414*, 813.
- (5) Hönes, J.; Müller, P.; Surridge, N. Diabetes Technol. Ther. 2008, 10, S10.
- (6) Minder, E.; Albrecht, D.; Schäfer, J.; Zulewski, H. *Diabetes Res. Clin. Pract.* 2013, 101, 57.
- (7) Lane, J. K.; Krumholz, D. M.; Sack, R. A.; Morris, C. Curr. Eye Res. 2006, 31, 895.
- (8) Baca, J. T.; Taormina, C. R.; Feingold, E.; Finegold, D. N.; Grabowski, J. J.; Asher, S. A. *Clin. Chem.* **2007**, *53*, 1370.
- (9) Taormina, C. R.; Baca, J. T.; Asher, S. A.; Grabowski, J. J. J. Am. Soc. Mass. Spec. 2007, 18, 332.
- (10) Cha, K. H.; Jensen, G. C.; Balijepalli, A. S.; Cohan, B. E.; Meyerhoff, M. E. Anal. Chem. 2014, 86, 1902.
- (11) Choy, C. K. M.; Benzie, I. F. F.; Cho, P. Invest. Ophthalmol. Vis. Sci. 2000, 41, 3293.
- (12) Choy, C. K. M.; Cho, P.; Chung, W. Y.; Benzie, I. F. F. Optom. Vis. Sci. 2003, 80, 632.
- (13) Yamaoka, H.; Sode, K. Open Biotechnol. J. 2007, 1, 26.
- (14) Tang, Z. P.; Louie, R. F.; Lee, J. H.; Lee, D. M.; Miller, E. E.; Kost, G.; J. Crit. Care Med. 2001, 29, 1062.
- (15) Wilson, R.; Turner, A. P. F. Biosens. Bioelectron. 1992, 7, 165.
- (16) Fapyane, D.; Lee, S. J.; Kang, S. H.; Lim, D. H.; Cho, K. K.; Nam, T. H.; Ahn, J. P; Ahn, J. H.; Kim, S. W.; Chang, I. S. *Phys. Chem. Chem. Phys.* **2013**, *15*, 9508.
- (17) Smith, L. D.; Budgen, N.; Bungard, S. J.; Danson, M. J.; Hough, D. W. *Biochem. J.* **1989**, *261*, 973.
- (18) Yuan, X.; Iijima, M.; Oishi, M.; Nagasaki, Y. Langmuir, 2008, 24, 6903.
- (19) Burke, D. W.; Marquant, M. Test Strip with Slot Vent Opening. U.S. Patent 8,211,379, July 3, **2012**.

CHAPTER 4

Determination of Correlation Between Tear and Blood Glucose Levels of Human Subjects with Type 2 Diabetes Using Roche Accu-Chek Test Strip*

4.1 Introduction

As summarized in the preceding two chapters, the Roche Accu-Chek blood glucose test strip is unique in its ability to measure the very low levels of glucose that are present in normal human subjects.¹⁻² The mechanism that enables these strips to accurately measure such low levels of glucose without significant interference from common redox species has also been examined/ determined.

Herein, we investigate the correlation between tear and blood glucose levels of Type 2 diabetic patients before and then over 1.5-h period after these patients consumed 50 g glucose in a tube of jelly. The measurements of the tear samples were made using the commercially available ACCU-CHEK Aviva Plus blood glucose test strips and a high sensitivity potentiostat, to definitively

^{*} The key experiments of this chapter were conducted by Kyoung Ha Cha, with help from Dr. Taj Khan, M.D. in providing diabetic patients.

establish whether or not tear glucose is a potential noninvasive supplemental method to blood glucose measurements.

4.2 Experimental Section

4.2.1 Materials

D-(+)-glucose, sodium chloride (NaCl), potassium chloride (KCl), sodium phosphate dibasic (Na₂HPO₄), potassium phosphate monobasic (KH₂PO₄), L-ascorbic acid, uric acid, and acetaminophen were purchased from Sigma-Aldrich (St. Louis, MO).

4.2.2 Sensitivity and Selectivity Analysis

To determine the sensitivity and selectivity of commercial glucometer strips, a four-channel Biostat (ESA Biosciences Inc., Chelmsford, MA) measured the amperometric response to glucose solutions (0-100 μ M) and to glucose solution (50 μ M) containing potential interfering species (easily oxidizable species at applied potentials). For testing of the commercial glucose strips, the reported applied potentials are versus the internal reference/counter electrode within the strips.

4.2.3 Human Tear Sample Acquisition and Analysis

Under an IRB approved protocol (through Rutgers University), fasting human subjects (n = 4) with type 2 diabetes were tested in New Jersey under the direction of Dr. Taj Khan, M.D. (ophthalmologist) who viewed the eyes of the human subjects, who were not anesthetized, through a slit lamp microscope at ×16 magnification. After separate tear samples were obtained from the lacrimal lake via hand-held 1- μ L disposable glass capillary tubes (micropipets "Microcaps" Drummond Scientific Co., Broomall, PA.) from the left and then the right eye of each subject, subsequent blood glucose levels were measured without intermission using a blood glucometer. A glucose challenge was introduced to the patients after the first set of samples were collected. This was accomplished by having the patients consuming 50 g glucose in a tube of jelly. From then, each subsequent set of samples were obtained with 30 min intervals for the next 90 min resulting in the total of 4 sets of tear collections per human subject. All human tear samples were measured using the same procedure described by Cha et al.¹⁻² with the ACCU-CHEK Aviva Plus test strips (Lot No. 496974, Roche Diagnostics, Indianapolis, IN). And all blood glucose levels were measured using the same strips with the paired commercial glucometer. For human tear measurements, 2 s average current values were recorded starting at the 2 s time-point after calibrant or sample introduction were used to prepare the calibration curve and quantitate tear values, respectively.

4.3 Results and Discussion

As shown in Figure 4.1a, a calibration of 0-100 μ M glucose obtained by using the Roche test strips (Lot No. 496722) yields a LOD and LOQ of 7 and 22 μ M (n = 3), respectively. The test strips also exhibit $\leq 6.7\%$ error (n = 3) for measuring 50 μ M glucose in the presence of potential interference species, including 10 μ M acetaminophen, 100 μ M ascorbic acid, and 100 μ M uric acid. As shown in Figure 4.1b, a different lot of the test strips (Lot No. 496974) with an LOD and LOQ of 2 and 7 μ M (n = 3) was evaluated to be used for human testing. Of note, calibration data for glucose used 2 s average current recorded starting at the 2-s time point after sample introduction with an applied potential of +150 mV. The method for averaging currents recorded is somewhat different from Chapter 2 but provides more accurate results. Error bars are SD for n = 3 replicate measurements. The reported average levels of ascorbic and uric acid in tears are 20 ± 6.2 and 68 ± 46 μ M,³²⁻³³ respectively. A 5 mM stock solution of glucose was made in phosphate saline buffer



Figure 4.1. Calibration data for glucose using 2 s average current recorded starting at the 2 s time point after sample introduction with an applied potential of +150 mV. (a) 0-100 μ M glucose calibration obtained by using test strips (Lot No. 496722). The limit of detection (LOD) and limit of quantitation (LOQ) are 7 and 22 μ M, respectively. The test strips exhibit $\leq 6.7\%$ error (n = 3) for 50 μ M glucose in the presence of potential interference including 10 μ M acetaminophen, 100 μ M ascorbic acid, and 100 μ M uric acid. (b) 0-400 μ M glucose calibration obtained by test strips (Lot No. 496974), used for human testing. The LOD and LOQ are 2 and 7 μ M, respectively. Error bars are SD for n = 3 replicate measurements.

(PBS) solution containing 137 mM NaCl, 2.7 mM KCl, 10 mM Na₂HPO₄, and 2 mM KH₂PO₄, adjusted pH to 7.4 with 1 M NaOH. Standard solutions were prepared by dilution of this stock solution using PBS solution for calibrations of glucose at 0, 50, and 100 μ M. Ten μ M acetaminophen was added to the test mixture, assuming that this species may be present in tears at a relative dilution ratio similar to blood glucose, if the patient was using this drug.

Tear samples obtained from n = 4 fasting, type 2 diabetic human subjects were collected in 1 μ L microcapillary tubes. As shown Figure 4.2, the chronological changes in tear and blood glucose levels of these 4 human subjects with type 2 diabetes clearly demonstrates a correlation. The



Figure 4.2. (a-d) Chronological changes in tear and blood glucose levels of n = 4 human subjects with type 2 diabetes. The collection order at every 0.5 h is as following: right eye, left eye and blood (drawn from lancet); Glucose challenge was introduced right after the first set of collection at 0-h mark. *data indicates a tear glucose measurement obtained from an irritated eye.

lowest level readings were for the samples obtained prior to the patients consuming 50 g glucose in a tube of jelly. The excluded value (marked with *) could have resulted from small perturbations induced by the subject rubbing his/her eye in advance of tear acquisition or by excess pressure applied to the surface of the eye during the tear acquisition using the micropipet. In both instances, tiny volumes of the higher glucose levels within tissue or blood capillaries within the eye lid could enter the tear fluid, temporarily increasing tear glucose values. It is also important to note that the same trend in change tear glucose levels can be observed in the both eyes of each human subject and that the magnitude of the tear glucose levels differ significantly for each patient.

As shown in Figure 4.3, the tear and blood glucose levels yield moderate correlation (R^2) values of 0.6294, 0.6174, 0.5642, and 0.7021 for each human subject with Type 2 diabetes that was tested,



Figure 4.3. (a-d) Correlation between tear and blood glucose levels in n = 4 human subjects with Type 2 diabetes.

respectively. Due to this study being a preliminary human study for tear and blood correlation and the very first of its kind using this type of glucose sensing method, the correlation values obtained are a reasonably good fit based on the number of human subjects tested. However, future studies are needed to improve the correlation value. In addition, such significant differences in the magnitude of tear glucose levels for each of the 4 patients suggests that it is imperative to further investigate what factors dictate the ratio between blood and tear glucose levels from patient to patient. Whether this ratio is also influenced by the patient having or not having diabetes also must be examined.

4.4 Conclusion

In this chapter, the ACCU-CHEK Aviva glucometer test strips were used to measure glucose levels in tear fluids from 4 fasting human subjects with Type 2 diabetes before and over a 1.5 h period after these patients consumed 50 g glucose in a tube of jelly. Blood glucose values were also measured as similar time points as the tear measurements. A moderate correlation between tear and blood glucose levels was found for each patient, suggesting that measurement of tear glucose is a potential noninvasive supplemental method for diabetic patients to monitor blood glucose levels. However, the difference in magnitude of tear glucose levels between patients suggests that a longer-term analysis of each patient (more than 1.5 h) may be the key to definitively establish whether the correlation between tear and blood glucose levels is good enough to be clinically useful for such patients.

4.5 References

- (1) Cha, K. H.; Jensen, G. C.; Balijepalli, A. S.; Cohan, B. E.; Meyerhoff, M. E. *Anal. Chem.* **2014**, *86*, 1902.
- (2) Cha, K. H.; Qin, Y.; Meyerhoff, M. E. *Electroanalysis* **2015**, *27*, 670.

CHAPTER 5

Compatibility of Nitric Oxide Release with Implantable Enzymatic Glucose Sensors Based on Osmium (III/II) Mediated Electrochemistry

Cha, K.H.; Meyerhoff, M.E. ACS. Sens. 2017; 2(9): 1262-1266 *

5.1 Introduction

Tight glycemic control is essential to prevent or reduce life-threatening hyper- and hypoglycemic events that can cause serious long-term complications for critically ill patients.¹ Therefore, the development of continuous glucose monitoring systems to quantitate blood glucose levels intravascularly (IV) would greatly enhance glycemic control and, ultimately, improve patient outcomes, especially within intensive care units.²⁻⁶ FDA approved electrochemical continuous glucose sensors (e.g., commercial Dexcom, Medtronic, and Abbott glucose sensor systems) rely on measurements being made within the subcutaneous interstitial fluid, not within the blood stream. In most cases, frequent calibrations (once or twice per day) are required, by using a blood glucometer test strip and

^{*} The key experiments of this chapter were conducted by Kyoung Ha Cha.

a finger-prick, to update the calibration between blood levels and the sensor's output current. However, it is known that a significant lag time exists between interstitial and blood glucose levels (5-10 min). Further, devices implanted subcutaneously can suffer from a foreign body response (FBR), which can "wall off" the sensor, altering its sensitivity and potentially increasing the lag time to detect changes in blood glucose levels.⁷⁻⁸ Therefore, for patients in a hospital setting, implantation of continuous glucose sensing devices within the bloodstream would be convenient and advantageous.

To date, such continuous IV glucose sensors have not been successfully developed due, in part, to activation of platelets that can adhere to and form blood clots on the surface of the sensor (see Chapter 1). The presence of adhered, metabolically active platelets can change (lower) the local glucose levels at the surface of the device. To overcome this issue, many efforts have been made to reduce cell adhesion on the surface of implanted sensors and other IV devices by using various types of modified polymers,⁹⁻¹⁴ yet none have been successful.

Nitric oxide (NO) is endogenously produced from endothelial cells that line the inner walls of all healthy blood vessels and is known to prevent platelet activation and subsequent thrombosis, as well as serve as a potent antimicrobial/anti-biofilm agent.¹⁵⁻¹⁸ Therefore, one potential strategy to prevent thrombosis and infection associated with intravascular sensing devices is to utilize polymer coatings that release NO at physiological levels of ca. $0.5 \sim 4.0 \times 10^{-10}$ mol·min⁻¹·cm⁻². Indeed, Yan et al. developed needle-type amperometric glucose sensors based on hydrogen peroxide detection with an NO releasing polymer and demonstrated that such sensors exhibit more accurate glucose measurements, as well as reduced thrombus formation over an 8 h period

compared to control sensors when implanted in the veins of rabbits.¹⁹ Soto et al. also demonstrated the potential of NO release to enhance the analytical performance of subcutaneous glucose biosensors.²⁰ Further, NO release coatings have also been successfully employed to improve the *in vivo* analytical accuracy of intravascular oxygen sensors.²¹⁻²² However, in the case of glucose sensors based conventional hydrogen peroxide detection (oxidation), devices with NO-releasing coatings that exhibit higher NO fluxes (desirable for more antithrombotic and antimicrobial activity) exhibit larger background currents in absence of glucose. This is due to the oxidation of NO at +0.65 V vs. Ag/AgCl at the surface of the underlying Pt/Ir working electrode of these sensors. Further, the level of this background current contribution decreases with time, as the NO flux decreases, thereby changing the low-end calibration curve for the sensor. Therefore, using a different glucose sensing method requiring a lower oxidation potential could further improve the compatibility of NO release coatings with implantable enzymatic glucose sensors.

In the 1990's, Heller and coworkers introduced the concept of a wired-enzyme electrode that is now utilized by Abbott Laboratory in their subcutaneous, continuous glucose-monitoring system.²³ Such an enzyme electrode is based on "wiring" glucose oxidase (GOX) to an electron-transfer osmium ion mediator, $Os(bpy)_2Cl^{+/2+}$ (osmium(II) complex with bipyridine), further complexed to poly(1-vinylimidazol) (PVI-Os). The potential required for oxidation of Os(II) to Os(III) is very low, <+400 mV vs. SCE, a potential that should avoid the additional background current from the oxidation of nitric oxide (to nitrate) at the inner working electrode, when NO release polymers are used to coat/contain the devices. Herein, we demonstrate the compatibility of nitric oxide with implantable enzymatic glucose sensors based on the osmium (III/II) chemistry by fabricating NO-releasing osmium-mediated glucose sensors and testing them in both phosphate buffer (pH 7.4) and *in vitro* within heparnized whole porcine blood at 37 °C. Amperometric glucose response is preserved in the presence of continuous NO release from the surrounding NO-releasing polymer, suggesting that NO does not interact in any way with the Os(III/II) redox center of Pt/Ir working electrode, and thus does not alter the amperometric signals of such glucose sensing devices.

5.2 Experimental Section

5.2.1 Sensor Fabrication

Two fabrication methods were employed to prepare the two types of needle-type glucose sensors employed in this work. One design (Fig. 5.1a) is based on the configuration reported by Bindra et al.,²⁴ Gifford et al.²⁵ and Yan et al.¹⁹ One μ L of 4.5 mg/mL PVI-Os (III) in methanol, 1 μ L of 0.9 mg/mL glucose oxidase (Type VII, from aspergillus niger) in a mix of methanol and HEPES (1:1 wt%), and 1 μ L of 0.3 mg/mL poly(ethylene glycol) diglycidyl ether (PEGDE) were dropped



Figure 5.1. Designs of implantable (a) osmium-mediated glucose sensor and (b) NO-releasing osmium-mediated glucose sensor.

sequentially into a 1 mm cavity created on the side of an insulated platinum-iridium wire (see Fig. 5.1a). The sensor was subsequently rinsed with water and allowed to dry for 1 h. Then, the sensor was dip-coated with a thin layer of Nafion to enhance selectivity over potential anionic interferences such as ascorbic and uric acid, and then allowed to dry overnight in a 37 °C oven. The sensor was then loop-coated with ca. 1% (wt/vol) PurSil polymer solution in THF and dried for 8 h to remove any THF via evaporation. To fabricate the NO-releasing osmium-mediated glucose sensor (Fig. 5.1b), a Pt-Ir wire insulated with a Teflon coating was inserted into a commercial silicone tubing (Artec) which was previously impregnated with the NO donor, Snitroso-N-acetylpenicillamine (SNAP), via a solvent swelling method using 125 mg/ml SNAP in THF.²⁸ Before insertion, the Teflon coating was coated with a layer of copper nanoparticles (Cu-NPs, 5 wt % and 40-60 nm) in Carbosil (2080A, DSM) in THF solution. Then, at the distal tip of the device, sequential layers of glucose oxidase, PVI-Os (III) and PEGDE, then Nafion and 2 % PurSil in THF were applied with drying in between each layer (see Fig 5.1b). The analogous control sensors were prepared by only inserting the osmium-mediated glucose sensor (with the sensing area at the distal tip) into the silicone tubing, but without the SNAP and Cu-NP layers.

5.2.2 Sensor Performance

Figure 5.2 shows that the basic osmium-mediated glucose sensors (Fig. 5.1a) without any NO-releasing outer tubing exhibit stable amperometric responses to glucose in the range of 0-20 mM using an applied potential of only +200 mV vs. Ag/AgCl between the Pt/Ir electrode and the wrapped Ag/AgCl reference electrode. As also shown, this response is quite stable for at least 5 days, when the sensor is stored and tested in phosphate buffer solution (PBS, pH 7.4) at 37 °C, with excellent linearity ($R^2 = 0.9963$) and reasonably fast response times (3-5 min). The sensitivity



Figure 5.2. Amperometric response to glucose (0-20 mM) obtained with non-NO release Os(III/II)-mediated implantable glucose sensor (Fig. 5.1a) with applied potential of 200 mV on days 1-5.

of glucose sensor slowly changed during storage in PBS at 37 °C over 5 consecutive days, which is typical behavior for devices prepared with a PurSil (PU-based) outer coating, due to swelling of PU layer due to water uptake during the initial hydration period period.^{24,26} The normalized sensitivity of the basic osmium-mediated glucose sensors (n = 3) on day 1, 3, and 5 day are as follows: 117.4 % and 118.4 % on Day 3 and 5, respectively. These values were calculated by dividing the sensitivity on each day by that on day 1.

5.3 Results and Discussion

5.3.1 Effect of Nitric Oxide on Osmium-Mediated Chemistry

To determine whether nitric oxide can cause additional background current to glucose sensors based on hydrogen peroxide detection (+0.65 V vs. Ag/AgCl reference), but not to the sensors based on osmium-mediated chemistry (+0.2 V), both types of glucose sensors were prepared without any NO-releasing outer tubing, and were placed in a bulk PBS solution (pH 7.4) with 0 mM glucose in the absence (purged with N₂) and the presence of 300 ppm NO. Notably, the concentration of a saturated NO solution is reported to be 1.74 ± 0.03 mM (equivalent to 52.2 ppm).²⁷ As shown in Figure 5.3, when 300 ppm NO was introduced at the 10 min mark, the



Figure 5.3. A comparison of continuous current measurements for 0 mM glucose using a glucose sensor based on hydrogen peroxide detection (+0.65 V vs. Ag/AgCl) and osmiummediated chemistry (+0.2 V) in the absence and presence of 300 ppm NO purged into the solution at the 10 min mark. Current was normalized by subtracting the background current at 0 mM glucose for both sensors.

osmium-mediated glucose sensor exhibits no additional background current while the sensor based on hydrogen peroxide detection exhibits a significant increase in current (~5 nA) (equivalent to 4.1 mM glucose based on prior calibration curve for that sensor) due to electrochemical oxidation of NO. This result confirms that the application of lower redox potential (+0.2 V) avoids the unnecessary additional background current from oxidation of NO at the inner electrode surface of sensing area.

To determine whether nitric oxide has any adverse effects on the electron-transfer process during glucose measurements, osmium-mediated glucose sensors prepared without any NO release outer tubing (Fig. 5.1a) were used to measure 14 mM glucose in bulk PBS solution (pH 7.4) while simultaneously purging the solution continuously with 300 ppm NO gas over a 25 h period. The osmium-mediated glucose sensor exhibits very stable and continuous measurement of 0 mM glucose (background current) between 0-3 h and 14 mM glucose between 3-25 h in both the absence (0-1 h) and presence (1-25 h) of 300 ppm NO (see Fig. 5.4). Furthermore, not only did NO fail to provide any additional background current at the applied potential of +200 mV (as



Figure 5.4. Continuous measurement of 0 mM glucose (0-3 h) and 14 mM glucose (3-25 h) using osmium-mediated glucose sensor with applied potential of 200 mV. Gas phase NO at 300 ppm in nitrogen was purged between 1-25 h. Inset: Corresponding calibration curve of same sensor for glucose (0-20 mM).

shown in Fig. 5.4 between 1-3 h), but the sensor also exhibited relatively stable and continuous measurement of the 14 mM glucose concentration in the presence of 300 ppm NO, more than the concentration of saturated NO in solution (52.2 ppm),²⁷ for 22 h (<4 % change in current over this period). These measurements match the calibration curve (N₂-purged solution) obtained prior to the experiment (inset in Fig. 5.4). As a result, the osmium-mediated glucose sensor exhibits continuous glucose sensing capability in the presence of highly concentrated levels of NO within the test solution.

5.3.2 Nitric Oxide Flux Analysis

As shown in Figure 5.5, the additional layer of Cu-NPs in Carbosil between the Teflon coating on the Pt working electrode and SNAP-impregnated silicone tubing increases the NO release flux. The Cu⁺ ions, produced from oxidation of Cu⁰ nanoparticles in the presence of moisture, catalyze the generation of NO from RSNOs such as SNAP.²⁹⁻³¹ Indeed, the SNAP- impregnated silicone tubing without the Cu-NP coatings exhibited lower NO fluxes compared to the Cu-NP based devices. The NO fluxes from SNAP with Cu-NPs correspond to the physiological levels (0.5-4 × 10⁻¹⁰ mol·min⁻¹·cm⁻²) of NO released by endothelial cells that line the walls of all blood vessels.³²



Figure 5.5. NO flux analysis of SNAP-based NO release glucose sensors prepared with and without Cu-NPs for 3 consecutive days (note: 1 flux unit = $1 \times 10^{-10} \text{ mol} \cdot \text{min}^{-1} \cdot \text{cm}^{-2}$). The error bars represent standard deviation for n = 3.

Nitric oxide released from the osmium-mediated glucose sensor was measured in real-time via chemiluminescence using a Sievers Nitric Oxide Analyzer (NOA) 280.

5.3.3 Detection of Copper Leaching

Detection of the copper ion leaching from the polymeric composites (e.g., Carbosil) is crucial for the preclinical testing of an implantable device, as diffusion of copper ions from the polymers will highly influence cell cytotoxicity. After 72 h of soaking in 8 mL of PBS (pH 7.4) at 37 °C to allow any potential copper ion leaching, the glucose sensors containing the Cu-NPs in the inner Carbosil layer were removed from the solution and a standard ICP-MS method was employed to monitor the degree of copper leaching from the sensors. The results of the detected copper leaching are shown in Table 5.1. Measured copper leachate from the sensors was well below cytotoxic concentrations for mammalian cells. Previous reports have indicated that cell viability stays at 100% until the concentration of copper reaches approximately 1000 ppb.³³⁻³⁵ The result demonstrates that the concentration of leached copper is insignificant and therefore should exhibit no adverse effect on sensor biocompatibility.

Table 5.1. Copper leachate concentration from Cu-NPs in Carbosil layer in osmium-mediated glucose sensor and relative leaching in PBS as compared to the original amount of copper present in each sensor

sample	initial Cu-NPs (µg)	measured copper leachate	% copper leaching
		in 8 mL PBS after 72 h (ppb)	after 72 h
Cu-NPs	19.05 ± 1.695	152.7 ± 4.967	6.45 ± 0.50
Cu-SNAP	18.25 ± 0.4490	74.59 ± 27.06	3.29 ± 1.25

5.3.4 Sensor Performance on Benchtop

As shown in Figure 5.6, although the NO-releasing glucose sensor exhibits slightly different glucose sensitivity than the control, this difference is not due to interference from NO, but is rather due to the fabrication process, as the amount of reagents (GOX, PVI-Os, and outer PU coating) deposited on sensing area directly dictates the current output for glucose measurements. Most importantly, continuous NO release from SNAP does not exhibit any negative effect on the osmium-mediated chemistry for glucose measurement over a 3-day period. The normalized sensitivity of NO-releasing and controls sensors (n = 3) over 3 days are as follows: 109.1 and 86.4 % on Day 2 and 3 for NO-releasing sensors, respectively; 87.7 and 97.7 % on Day 2 and 3 for control sensors, respectively. The percentage was calculated from dividing the sensitivity on each



Figure 5.6. Glucose calibrations (0-20 mM) with applied potential of 200 mV in PBS (pH 7.4, 37 °C) for 3 consecutive days. (a) NO-releasing glucose sensor; (b) control glucose sensor. (note: 1 flux unit = $1 \times 10^{-10} \text{ mol·min}^{-1} \cdot \text{cm}^{-2}$).



Figure 5.7. (a-c) *In vitro* comparison of intermittent glucose values obtained from a commercial glucometer and continuous glucose measurements obtained from a NO-releasing osmium-mediated glucose sensor as well as a control (No NO release) osmium-mediated glucose sensor poised at +200 mV vs. Ag/AgCl in heparinized whole porcine blood for 3 days. On each day, glucose values were back-calculated using calibration data obtained prior to the in vitro blood testing.

day by that on Day 1. *In vitro* continuous glucose measurements in heparinized whole porcine blood (female, 120 kg, provided by Extracorporeal Life Support Laboratory at University of Michigan) was conducted with NO-releasing and control osmium-mediated glucose sensors for 3 consecutive days. On each day, glucose values were back-calculated using the calibration obtained prior to the *in vitro* blood testing. To confirm the accuracy of the glucose sensors, a commercial blood glucometer (Accu-Chek[®] Nano, Roche Diabetes Care) was used every 10 min to measure (in triplicate) the whole blood glucose level. This value was compared to the value obtained from the sensors using the prior calibration data. NO release was monitored for the three days via collecting NOA data for one hour each day, prior to glucose calibration (0-20 mM) in PBS (pH 7.4, 37 °C) and the *in vitro* continuous glucose measurements in porcine blood. The sensors were

stored in PBS (pH 7.4) at 37 °C in the dark between measurements to ensure continuous NO release from SNAP.

As shown in Figure 5.7, the intermittent glucometer glucose measurement values in heparinized whole porcine blood are very comparable to the continuous glucose measurements obtained by the osmium-mediated glucose sensors with the pre-calibration made in PBS on each day. The discrepancies in glucose value between NO-releasing glucose sensor and glucometer observed at the 2 min mark may be have resulted from glucometer test strips having the standard accuracy of $\pm 15\%$ for measured glucose concentration. Regardless, it is clear that during 3 days of continuous NO release from the sensors, at or above physiological levels, no negative effects on the osmium-mediated wired enzyme electrode chemistry are observed.

5.3.5. Improving Sensors to Reduce Oxygen Sensitivity

The NO release osmium-mediated glucose sensors were further optimized to avoid interference from oxygen by switching the glucose-oxidizing enzyme from glucose oxidase (GOX) to glucose dehydrogenase (GDH). The GDH utilizes flavin adenine dinucleotide (FAD) as a cofactor within the structure of the enzyme. Indeed, GOX can utilize both PVI-Os and oxygen as the electron acceptor which may generate unintended errors due to variations in the oxygen concentration/saturation of the blood (e.g., higher oxygen levels reduce the signal). However, FAD-GDHs are unable to utilize oxygen as an electron acceptor, despite possessing significant structural similarities with GOX.³⁶ As shown in Figure 5.8 (left), osmium-mediated glucose sensors that now utilize FAD-GDH as the glucose-oxidizing enzyme exhibit excellent linear sensitivity to glucose and selectivity over possible interference species such as 0.5 mM ascorbic



Figure 5.8. (Left) Glucose calibrations (0-20 mM) and obtained by new osmium-mediated glucose sensors (n=4) that utilize FAD-GDH as glucose-oxidizing enzyme with applied potential of 200 mV in PBS (pH 7.4, 37 °C). (Right) Continuous measurement of 20 mM glucose (0-24 h) using the new osmium-mediated glucose sensors (n=3) with applied potential of 200 mV in the presence of gas phase NO at 100 ppm in nitrogen. The electronic noise recorded after 84-min mark is from bubbling of N₂ into the solution.

acid (AA) and 0.2 mM acetaminophen (AC). And the current response to 20 mM glucose is not altered before and after purging the sample solution with nitrogen gas at the 84 min mark. This suggests that the PVI-Os is the only electron acceptor utilized by FAD-GDH with an applied potential of +200 mV vs. Ag/AgCl. Furthermore, the new sensor design with FAD-GDH exhibited relatively stable and continuous measurement of the 20 mM glucose in the presence of gas phase NO at 100 ppm nitrogen, more than the concentration of saturated NO in solution (52.2 ppm),²⁷ for 24 h (<4 % change in current over this period). As a result, the new osmium-mediated glucose sensor utilizing FAD-GDH as glucose-oxidizing enzyme exhibits continuous glucose sensing capability in the presence of highly concentrated levels of NO within the test solution.

5.3.6 Sensor Performance in Vivo

These new FAD-GDH based NO release glucose sensors were employed in a porcine model (~ 50 kg, n = 2 pigs) to determine their accuracy in monitoring blood glucose levels continuously *in vivo*. The sensors were placed in the femoral and carotid arteries of a pig under anesthesia, without any



Figure 5.9. Performance of electrochemical NO-releasing osmium-mediated glucose sensors implanted in pig arteries for 19 h at +200 mV vs. Ag/AgCl. *In vivo* comparison of intermittent glucose values obtained from a benchtop blood gas analyzer (red circles) and the converted current values measured by (a) two control sensors and (b) two NO-releasing sensors 1-point calibration was applied to data obtained by NO-releasing sensors at 10 h time point. (c) Representative photo illustrating the degree of clot formation on the surface of controls and the NO-releasing sensors after being explanted. (d) Average clot formation area on controls vs. NO-releasing sensors (n = 3, p < 0.05).

anti-coagulant administration, via an open cut-down allowing for continuous blood flow past the sensors for 19 h. The *in vivo* glucose level was measured continuously and changed periodically with 20 g of dextrose injections (ca. every 3 h). Arterial blood was drawn at 10 min, 20, 30, 60, 90, 120 and 180 min mark after every dextrose injection to assess the accuracy of the glucose

values provided by the implanted sensors. At the end of the *in vivo* experiments, the sensors with the blood vessels intact were explanted after systemic heparinization to prevent necrotic thrombosis during vessel harvesting; digital pictures were taken, and the red pixels were counted using ImageJ software to quantify clot area.³⁷⁻³⁸ The *in vivo* data from the sensors were recorded every second and averaged every 30 s to reduce the electronic noise as well as the size of the data set. A 1-point calibration was applied at the 10 h-point to further correct for the slight change of inherent sensitivity of glucose sensors with time (see Figure 5.6). As shown in Figure 5.9 (a) and (b), the NO-releasing sensors exhibited more reliable continuous glucose measurement for the entire 19 h in vivo experiments while the control sensors exhibited a significant deviation at almost every time point after 6 h of implantation. This was likely due to the formation of the large thrombus formation on the control sensor (see Fig. 5.9c, as an example). Thrombus formation around the sensing area can prevent direct contact with blood for glucose measurement. Overall, the NO releasing sensors induced less clot formation than the control sensors as quantitated by imaging the surface of the sensors following explantation from the pig after 19 h (see Fig. 5.9d, n = 3 for each type of sensor). Similar analytical results for most of the control and NO release sensors were obtained in the second animal study that was conducted with these new NO release in vivo glucose sensors (data not shown).

5.4 Conclusion

In summary, it has been shown that *in vivo* type amperometric glucose sensors prepared with the Os(III/II)-bipyridine-PVI complex maintain continuous analytical functionality and measurement accuracy in the presence of nitric oxide. Further, there is no evidence that NO can be oxidized at the underlying Pt/Ir working electrode at the applied potential of +200 mV vs. Ag/AgCl used to

record the current response from these devices. In addition, there does not appear to be any direct interaction between the NO and the Os(III/II) redox center that inhibits electron transfer from the glucose oxidase enzyme and this redox site. Similar behavior was observed when glucose oxidase or FAD-GDH was used as the wire enzyme of the sensors. The FAD-GDH system was found to not exhibit any oxygen sensitivity making it more useful for continuous *in vivo* glucose sensing. Preliminary data on the performance of these sensors in a pig model over a 19 h period are promising in terms of glucose sensing accuracy. While the ultimate focus of this research was to prepare useful IV glucose sensors, incorporation of NO release also has the potential to improve the performance (reduce drift, require less frequent calibrations) of similar wired enzyme electrodes employed for subcutaneous glucose sensing. This is because it is well known that NO release can also greatly reduce the foreign body inflammatory response that occurs for devices placed in the subcutaneous space.³⁹⁻⁴⁰

5.5 References

- (1) Hönes, J.; Müller, P.; Surridge, N. *Diabetes Technol. Ther.* **2008**, *10*, S10.
- (2) Van Den Berghe, G.; Wouters, P.;Weekers, F.;Verwaest ,C.; Bruyninckx, F; Schetz, M.; Vlasselaers, D.; Ferdinande, P; Lauwers, P.; Bouillon, R. N. Engl. J. Med. 2001, 345(19), 1359.
- (3) Bochicchio, G. V.; Scalea, T. M. Serono. Sym. 2008, 42, 261.
- (4) Kondepati, V. R.; Heise, H. M. Anal. Bioanal. Chem. 2007, 388(3), 545.
- (5) Schetz, M.; Vanhorebeek, I.; Wouters, P. J.; Wilmer, A.; van den Berghe, G. J. Am. Soc. Nephrol. 2008, 19(3), 571.
- (6) Vanhorebeek, I.; Langouche, L.; Van den Berghe, G. Chest 2007, 132(1), 268.
- (7) Sharkawy, A. A.; Klitzman, B.; Truskey, G. A.; Reichert, W. M. J. Biomed. Ma-ter. Res. **1997**, *37(3)*, 401.
- (8) Jablecki, M.; Gough, D. A. Anal. Chem. 2000, 72(8), 1853.
- (9) Kolobow, T.; Stool, E. W.; Weathersby, P. K.; Pierce, J.; Hayano, F.; Suaudeau, J. *Trans. Am. Soc. Artif. Intern. Organs* **1974**, *20A*, 269.
- (10) Szycher, M. J. Biomater. Appl. 1988, 3, 297.
- (11) Edmunds, Jr. L. H. ASAIO J. **1995**, 41, 824.
- (12) Didisheim, P. ASAIO J. **1994**, 40(2), 230.
- (13) Kim, W. S.; Jacobs, H. Blood Purif. 1996, 14(5), 357.

- (14) Larm, O.; Larsson, R.; Olsson, P. *Biomater. Med. Devices, Artif. Organs* **1983**, *11(2-3)*, 161.
- (15) Isenberg, J. S.; Romeo, M. J.; Yu, C.; Yu, C. K.; Nghiem, K.; Monsale, J.; Rick, M. E.; Wink, D. A.; Frazier W. A.; Roberts, D. D. *Blood* **2008**, *111(2)*, 613.
- (16) Nguyen, B. L.; Saitoh, M.; Ware, J. A. Am. J. Physiol. 1991, 261, H1043.
- (17) Sato, Y.; Hiramatsu, Y.; Homma, S.; Sato, M.; Sato, S.; Endo S.; Sohara, Y. J. Thorac. *Cardiovasc. Surg.* **2005**, *130(2)*, 346.
- (18) Zimmermann, A. K.; Aebert, H.; Reiz, A.; Freitag, M.; Husseini, M.; Ziemer G.; Wendel, H. P. ASAIO J. 2004, 50(3), 193.
- (19) Yan, Q.; Major, T. C.; Bartlett, R. H.; Meyerhoff, M. E. *Biosens. Bioelectron.* **2011**, *26(11)*, 4276.
- (20) Soto, R. J.; Privett, B. J.; Schoenfisch, M. H. Anal. Chem. 2014, 86(14), 7141.
- (21) Wu, Y.; Rojas, A. P.; Griffith, G. W.; Skrzypchak, A. M.; Lafayette, N.; Bartlett, R. H.; Meyerhoff, M. E. Sens. Actuators B Chem. 2007, 121(1), 36.
- (22) Ren, H.; Coughlin, M. A.; Major, T. C.; Aiello, S.; Rojas Pena, A.; Bartlett, R. H.; Meyerhoff, M. E. Anal. Chem. 2015, 87, 8067.
- (23) Pishko, M. V.; Michael, A. C.; Heller, A. Anal. Chem. 1991, 63(20), 2268.
- (24) Bindra, D. S.; Zhang, Y. N.; Wilson, G. S.; Sternberg, R.; Thevenot, D. R.; Moatti, D., Reach, G. Anal. Chem. 1991, 63(17), 1692.
- (25) Gifford, R.; Batchelor, M. M.; Lee, Y.; Gokulrangan, G.; Meyerhoff, M. E.; Wilson, G. S. *J. Biomed. Mater. Res.* **2005**, *75(4)*, 755.
- (26) Vaughn, M. W.; Kuo, L.; Liao, J. C. Am. J. Physiol. 1998, 274, H2163.
- (27) Mesároš, Š.; Grunfeld, S.; Mesárošová, A.; Bustin, D.; Malinski, T. Analytica Chimica Acta 1997, 339(3), 265.
- (28) Brisbois, E. J.; Major, T. C.; Goudie, M. J.; Bartlett, R. H.; Meyerhoff, M. E.; Handa, H. *Acta Biomaterialia*, **2016**, *37*, 111,
- (29) Major, T. C.; Brant, D. O.; Burney, C. P.; Amoako, K. A.; Annich, G. M.; Meyerhoff, M. E.; Handa, H.; Bartlett, R. H. *Biomaterials* **2011**, *32(26)*, 5957.
- (30) Williams, D. L. H. Chem. Commun. 1996, 1085.
- (31) Pant, J.; Goudie, M. J.; Hopkins, S. P.; Brisbois, E. J.; Handa, H. ACS Appl. Mater. Interfaces 2017, 9(18), 15254.
- (32) Chen, X.; Matsumoto, N.; Hu, Y.; Wilson, G. S. Anal. Chem. 2002, 74, 368.
- (33) Fahmy, B.; Cormier, S. A. Toxicol. In Vitro 2009, 23, 1365.
- (34) Seth, R.; Yang, S.; Choi, S.; Sabean, M.; Roberts, E. Toxicol. In Vitro 2004, 18, 501.
- (35) Valodkar, M.; Rathore, P. S.; Jadeja, R. N.; Thounaojam, M.; Devkar, R. V.; Thakore, S. *J. Hazard. Mater.* 2012, 201-202, 244.
- (36) Ferri, S.; Kojima, K.; Sode, K. J. Diabetes Sci. Technol. 2011, 5, 1068.
- (37) Ren, H.; Colletta, A.; Koley, D.; Wu, J.; Xi, C.; Major, T. C.; Bartlett, R. H.; Meyerhoff, M. E. *Bioelectrochemistry* 2015, 104, 10.
- (38) Schneider, C. A.; Rasband, W. S.; Eliceiri, K. W. Nat. Methods 2012, 9, 671.
- (39) Hetrick, E. M.; Prichard, H. L.; Klitzman, B.; Schoenfisch, M. H. *Biomaterials* **2007**, *28(31)*, 4571.
- (40) Nichols, S. P.; Koh, A.; Brown, N. L.; Rose, M. B.; Sun, B.; Slomberg, D. L.; Riccio, D. A.; Klitzman, B.; Schoenfisch, M. H. *Biomaterials* 2012, *33*(27), 6305.

CHAPTER 6 Conclusions and Future Directions

6.1 Conclusions

Diabetes mellitus has become a leading cause of death worldwide. To date, although there is no ultimate cure to prevent or reverse diabetes, only targeted administration of appropriate medication and use of various blood glucose monitoring systems to enhance treatment efficiency, can help alleviate the symptoms and diminish the complications of the condition. This dissertation aimed to address and improve both a non-invasive and a continuous *in vivo* approach to glucose monitoring and thereby improve the measurement technology, potential patience compliance, and ultimately, patient outcome.

In Chapter 2, test-selected lots of commercially available ACCU-CHEK® Aviva Plus glucometer strips manufactured by Roche were demonstrated to exhibit the sensitivity and selectivity necessary to measure low levels of glucose in human tears using sample volumes of $<1 \mu$ L. Preliminary measurements of tear glucose using these strips in non-diabetic volunteers match well with those reported by Asher¹ with research laboratory LC-MS instrumentation.

In Chapter 3, it was found that that the ability of Roche Accu-Chek test strips to be used to measure μ M levels of glucose without substantial interference from other redox active species is specifically due to the unique combination of using PQQ-GDH as the active enzyme in combination with a nitrosoaniline derivative as an electron transfer mediator. The low K_M value of this enzyme, coupled with the inability of uric acid and acetaminophen to reduce the nitrosoaniline mediator and yield electrochemical response in a biamperometric measurement mode, appear to be the key elements that provide the excellent LOQ value and selectivity of these strips.

In Chapter 4, the ACCU-CHEK Aviva glucometer test strips were used to measure glucose levels in tear fluids from n=4 human subject with Type 2 diabetes over 1.5 h period while also concurrently measuring the blood glucose values. Measurements of blood and tear (in both eyes) glucose levels were made immediately before having the subjects consumed a tube of sugar jelly, and then for 1.5 h thereafter at 30 min intervals. A moderate correlation between tear and blood glucose levels was found for these diabetic patients, suggesting that measurement of tear glucose is a potential noninvasive supplement method for monitoring blood glucose levels. However, it appears that the ratio of glucose levels in tear fluid vs. blood is variable from patient to patient, and hence this approach may require calibration of this ratio for each patient.

In Chapter 5, *in vivo* type amperometric glucose sensors prepared with the Os(III/II)-bipyridine-PVI complex were shown to maintain continuous analytical functionality and measurement accuracy in the presence of nitric oxide release. Further, there is no evidence that NO is oxidized at the underlying Pt/Ir working electrode at the applied operational potential of +200 mV vs. Ag/AgCl reference electrode used to record the current response from these devices. In addition, there does not appear to be any direct interaction between the NO and the Os(III/II) redox center that inhibits electron transfer from the glucose oxidase enzyme and this redox site. By using FAD-glucose dehydrogenase (FAD-GDH) in place of GOX as the glucose selective enzyme, there was no interference from variable levels of oxygen in the sample phase, and this would be quite advantageous for use of such sensors for continuous IV monitoring of glucose in patients within ICU or CC units within hospitals.

Overall, this thesis work demonstrated significant potential improvements in both non-invasive and continuous glucose monitoring systems. The tear glucose measurement approach requires more extensive clinical studies to definitively establish the true correlation between tear and blood glucose levels, and how variable the ratio of these values are from patient to patient. The continuous glucose sensing devices using osmium-mediated chemistry and SNAP-doped silicone tubing aim to bridge the divide between the basic research on continuous monitoring glucose electrochemical sensors and a more clinically viable commercial product. Indeed, the use of wired FAD-GDH to ultimately develop an improved NO releasing intravascular glucose sensor that can provide more accurate, reproducible, and continuous real-time glucose measurements for critically ill hospital patients without risk of clotting (or infection; owing to NO's potent antimicrobial activity) seems like a very promising new approach.

6.2 Future Directions

6.2.1 Tear Fluid Acquisition Method for Tear Glucose Measurement

The future clinical potential for diabetes management of tear glucose testing with glucometer strips requires thorough evaluation in human subjects, including responses to the full range of blood glucose levels during hyperglycemia and, especially, hypoglycemia. The practicality of self-acquisition of tear sample must also be addressed for safety and efficacy before such a method could be clinically employed.

Collecting large volumes of tear fluid (>1 μ L) can cause a significant amount of irritation and induce stress on the ocular surface. Leakage of intercellular fluid from damaged conjunctiva can occur and consequently increase measured tear glucose levels. Elevated tear glucose levels can be prolonged for up to 30 minutes, further skewing the correlation with blood glucose.² Therefore, to accurately detect tear glucose concentration, it is important to minimize the collection volume to sub-microliters and ensure no eye irritation occurs throughout the process.

Tear production via non-stimulation is the most reliable method for tear fluid collection which typically utilizes capillary tubes to collect ~1 μ L of tear fluid by gently touching the tear film on ocular surface.³ However, given the safety concern of holding a small tube on a subject's eye, such tests would have to be done by trained clinical personnel (as performed in Chapter 4) adding additional challenge to collection methods. Moderate correlation between tear and blood glucose levels was observed but variation within individuals and uncontrolled irritation still is challenging



Figure 6.1. Potential design of tear glucose sensor featuring a soft housing for holding the absorptive material that may sit directly above the sensing area.³

the viability of using tear glucose measurement for optimal glycemic control among diabetic patients.

Requirements for an operational tear glucose sensor device must include the ability to collect a low volume of fluid and cause no irritation to the eye. Indeed, Labelle et al. proposed a prototypical fluid capture device that would utilize a soft foam plug to collect a small amount of tear fluid (see Figure 6.1).⁴ Their potential design aims to provide a supplementary method to blood glucose monitoring system that is non-invasive, rapid, and safe, which may ultimately increase patient compliance. Testing of such a device in conjunction with the glucometer strip measurement of tear glucose would be a logical next step for future research in this area.

6.2.2 Development of Insulin Delivery Cannula with Built-In Osmium-mediated Glucose Sensor A combination of continuous glucose monitoring system with continuous subcutaneous insulin infusion therapy has led to significant reduction in hemoglobin A1c(HbA1c) in subjects with type 1 diabetes.⁵ Therefore, a single device that delivers both insulin infusion and continuous subcutaneous glucose monitoring would likely be an optimal strategy to manage changing glucose levels of type 1 diabetic patients, considering that insulin infusion therapy causes insignificant



Figure 6.2. Glucose-sensing cannula results obtained in anesthetized nondiabetic swine using both H_2O_2 -detection electrode (upper figure) and redox mediator sensor (lower figure). Arrows at 0 min time-marks in both upper and lower figures represent the timing of lispro insulin delivery. Note significantly high spurious current response to the insulin formulation in upper figure (from ref. 5).

perturbation to subcutaneous glucose levels at sites adjacent (e.g., 0.9 cm) to insulin infusion location.

Indeed, Ward et al.⁶ evaluated the performance of both H_2O_2 -detection and redox mediator glucose sensors in anesthetized nondiabetic swine for potential interferences in insulin formulations and reported that phenolic preservatives present in insulin formulations can cause significant interference in glucose sensing by being oxidized at high redox potential (+0.6-0.65 V) used by H_2O_2 -detection method. However, this problem can be avoided by using a redox mediator chemistry, such as osmium-mediated chemistry, that requires much lower redox potential of +0.175 V vs. Ag/AgCl (as shown in Fig. 6.2). Therefore, the research reported in Chapter 5 of



Figure 6.3. Calibration curves obtained using glucose sensors based on H_2O_2 -detection (left) and NO fluxes (right) of the integrated device in a 18-day experiment in phosphate buffer solution.

this thesis on osmium-mediated continuous glucose sensors that uses a redox potential of +0.15 V, would likely be a perfect candidate to be coupled with insulin infusion therapy for an ultimate bifunctional infusion cannula/glucose sensor system.

Indeed, ongoing research by Dr. Xuewei Wang in this lab has focused on developing such insulin delivery cannula system integrated with the osmium-mediated glucose sensor that is immune to phenolic preservatives present in insulin formulations. Furthermore, nitric oxide, a potent vasodilatory, anti-inflammatory, antimicrobial/bactericidal, and pro-angiogenesis agent can be utilized to mitigate foreign body response of such a subcutaneous device to have an extended life time (e.g., 2 to 4 weeks). In his preliminary work (Fig. 6.3), the glucose sensors based on H₂O₂-detection exhibits stable sensitivity to glucose levels between 0 to 20 mM in phosphate buffer solution and releases nitric oxide > 0.5 x 10^{-10} mol-cm⁻²-min⁻¹ flux from SNAP-impregnated cannula for 18 days. This cannula system is now in the process of being coupled with osmium-mediated glucose sensor, instead of H₂O₂-detection method, to evaluate its performance in the presence of phenolic preservatives in insulin formulations.



Figure 6.4. (a) Calibration of 3^{rd} generation glucose sensor utilizing CDH in TRIS buffer (pH 7.4) with an applied potential of +0.25 V vs. Ag/AgCl. Inset: Linear part of the calibration curve. (b) Biosensor response over the pH range (4.5-8). (c) Biosensor response over the temperature range (20-40 °C). (d) Lifetime of the biosensor in the presence of 0.75 mM glucose solution. (e) Interference on glucose response in presence of 0.75 mM glucose, maltose, cortisol, ascorbic acid, Ca²⁺ (from ref. 10).

6.2.3 Development of NO-releasing 3rd Generation Glucose Sensor

3rd Generation biosensors employ direct electron transfer between the electrode and the active site of the enzymes, with a low redox potential of the enzyme itself, eliminating the need of toxic and artificial electron mediators, thereby avoiding errors caused by different concentration of oxygen

in blood.⁷ Unfortunately, only a few enzymes are able for direct electron transfer between their active sites to the elctrode.⁸⁻⁹ Cellobiose dehydrogenase (CDH) has been received great attention as the active enzyme for 3rd generation biosensors due to its ability to directly transfer electrons with the electrode. Therefore, CDH would be a great candidate to investigate for the compatibility with nitric oxide release, to improve biocompatibility of a potential intravascular 3rd generation biosensors.

Indeed Bollella et al.¹⁰ recently reported that CDH and a novel glassy carbon electrode modified with direction electrodeposition of gold nanoparticles are capable of creating efficient direct electron transfer to be useful as a glucose biosensor. As shown in Figure 6.4, the biosensors are able to yield a low detection limit toward glucose of 6.2 μ M, a great linearity (R² = 0.995) in the wide range between 0.02 and 30 mM with an applied potential of +0.25 V vs. Ag/AgCl, which could be advantageous in selectivity over potential interferents such as ascorbic acid, uric acid and acetaminophen. Therefore, it would be interesting to investigate the compatibility of CDH with NO release for improving the biocompatibility of potentially implantable glucose biosensors.

6.3 References

- (1) Baca, J. T.; Taormina, C. R.; Feingold, E.; Finegold, D. N.; Grabowski, J. J.; Asher, S. A. *Clin. Chem.* **2007**, *53*, 1370.
- (2) Daum, K.; Hill, R. Acta Ophthalmol. 1984, 62, 530.
- (3) Baca, J. T.; Finegold, D. N.; Asher, S. A. *The Ocular Surface*, 2007, 5, 280.
- (4) La Belle, J. T.; Adams, A.; Lin, E; Engelschall, E.; Pratt, B.; Cook, C. B. *Chem. Commun.* **2016**, *52*, 9197.
- (5) Bergenstal, R. M.; Tamborlane, W. V.; Ahmann A, N. Engl. J. Med. 2010, 363, 311.
- (6) Ward, W. K.; Heinrich, G.; Breen, M.; Benware, S.; Vollum, N.; Morris, K.; Knutsen, C.; Kowalski, J. D.; Campbell, S.; Biehler, J.; Vreeke, M. S.; Vanderwerf, S. M.; Castle, J. R.; Cargill, R. S. *Diabetes Technol. Ther.* **2017**, *19*, 226.
- (7) Mazzei, F.; Favero, G.; Bollella, P.; Tortolini, C.; Mannina, L.; Conti, M. E.; Antiochia, R. *Int. J. Environ. Health* **2015**, *7*, 267.
- (8) Matsumura, H.; Ortiz, R.; Ludwig, R.; Igarashi, K.; Samejima, M.; Gorton, L. *Langmuir* **2012**, *28*, 10925.
- (9) Ludwig, R.; Ortiz, R.; Schulz, C.; Harreither, W.; Sygmund, C.; Gorton, L. *Anal. Bioanal. Chem.* **2013**, *405*, 3637.
- (10) Bollela, P.; Gorton, L.; Ludwig, R.; Antiochia, R. Sensors 2017, 17, 1912